# **ŤU**Delft

### **Bachelor Thesis**

### Transcutaneous Vagus Nerve Stimulation Through an In-Ear Device For Epilepsy Treatment

By,

K. Broek J. Teurlings B. Zuidema

July 4, 2022

Delft University of Technology

FACULTY OF ELECTRICAL ENGINEERING, MATHEMATICS AND COMPUTER SCIENCE

ELECTRICAL ENGINEERING PROGRAMME

## Abstract

Several studies have illustrated that Transcutaneous Vagus Nerve Stimulation (t-VNS) is an effective method for epileptic treatment. In contrast to the classical invasive Vagus Nerve Stimulation (VNS) which is already a Food and Drug Administration (FDA) approved therapy for epileptic treatment, t-VNS does not require surgical intervention. However, t-VNS has still not been evolved into an application which can be used in clinical practise due to a lack of power efficiency or size constraints. In this bachelor graduation project, an electrical circuit has been designed and implemented for stimulating the auricular branch of the vagus nerve. This circuit is part of a larger project where an ear piece is developed which is able to process Electroencephalogram (EEG) signals as well. The main design goals of the stimulator were power efficiency and safety. Furthermore, a design for an integrated circuit has been proposed resulting in an power efficient method for controlling the electrical stimulation circuit. The integrated circuit has, among other things, built in safety features and closed loop stimulation capabilities. Simulations showed that the designed circuit is able to successfully stimulate with current amplitudes up to 8 mA when taking worst case values for an impedance model of the electrode-skin interface. However, the implemented physical design was planned to be tested after the thesis submission. In other words, this thesis only contains the design and simulation phase of the stimulator. The design was found to be safe and power-efficient and could lay the basis for future development in the field of wearable in-ear stimulation devices.

## **Abbreviations**

ABVN Auricular Branch of the Vagus Nerve ADC Analog to Digital Converter AED Anti-Epileptic Drugs ASIC Application Specific Integrated Circuit BJT Bipolar Junction transistor DAC Digital to Analog Converter DC Direct Current DRE Drug Resistant Epilepsy EEG Electroencephalogram FDA Food and Drug Administration FPGA Field Programmable Gate Array FSM Finite State Machine **GND** Ground WHO World Health Organization

<b>MOSFET</b> Metal-Oxide-Semiconductor Field- Effect Transistor
NTS Solitary Nucleus
OCI Overcurrent Indicator
PCB Printed Circuit Board
<b>PNS</b> Peripheral Nerve Stimulation
<b>PWM</b> Pulse-Width Modulation
SUDEP Sudden Unexpected Death Epilepsy
t-VNS Transcutaneous Vagus Nerve Stimulation
TENS Transcutaneous Electrical Nerve Stimula- tion
UCI Undercurrent Indicator
VHDL VHSIC Hardware Description Language
<b>VN</b> Vagus Nerve
<b>VNS</b> Vagus Nerve Stimulation

IC Integrated Circuit

LC Locus Coeruleus

LDO Linear Dropout regulator

# Contents

Ał	breviations	2
1	Introduction         1.1       Epilepsy and Current Treatments         1.2       State of the art for t-VNS         1.3       Overview of the whole project         1.4       Document lay-out	<b>5</b> 5 6 6 6
2	Background         2.1 Impedance of the Skin         2.2 Electrodes         2.3 The electrode-skin model for the testing ground         2.4 Wave Characteristics	7 7 7 8 10
3	Programme of Requirements         3.1       Functional Requirements         3.2       Non-Functional Requirements         3.3       Requirement Explanation	<b>12</b> 12 12 13
4	Design Overview	14
5	Circuit Design         5.1       Stimulation Source         5.1.1       Current Source Implementations         5.1.2       Results         5.2       Current Sense         5.2.1       Current Sense implementations         5.2.2       Simulation Results         5.3       H-bridge         5.3.1       H-Bridge Implementations         5.3.2       H-Bridge Implementations         5.3.3       H-Bridge Simulations         5.4       Buck Converter         5.5       Boost Converter Implementations         5.5.1       Boost Converter Implementations         5.5.2       Simulation results	<ol> <li>16</li> <li>16</li> <li>18</li> <li>19</li> <li>20</li> <li>21</li> <li>21</li> <li>22</li> <li>23</li> <li>24</li> <li>24</li> <li>25</li> <li>26</li> </ol>
6	Integrated Circuit         6.1       Safety         6.2       Efficiency	<b>27</b> 28 28

#### CONTENTS

7	Discussion of the Results         7.1       Simulated Results         7.2       Prototype Results	<b>29</b> 29 29
8	Prototype implementation8.1PCB DesignPCB Design	<b>31</b> 31
9	Conclusion	32
10	Recommendations & Future Work	33
	Appendices	33
A	Background         A.1       Project Overview         A.2       Impedance Model & Electrodes	<b>34</b> 34 34
B	Circuit Design         B.1       Stimulation Source         B.2       H-Bridge         B.3       Current Sense         B.4       Buck Converter         B.5       Boost Converter         B.5.1       Component Values Calculation         B.5.2       Simulation results	<b>38</b> 38 39 40 41 42 42 43
С	Integrated Circuit         C.1       Decoder         C.2       H-bridge Control         C.3       Current Sense Control         C.4       DAC         C.5       Boost Control         C.6       Main Control         C.7       Signal Declarations	<b>48</b> 48 50 53 54 56 59 61

4

# Chapter 1 Introduction

### 1.1 Epilepsy and Current Treatments

Epilepsy is a neurological disorder that causes the brain to act abnormally. Epilepsy causes as a result of these recurring epileptic seizures. These seizures can range from mild to severe, in some cases resulting in physical injury. If left untreated, seizures can occur in sleep which could result in Sudden Unexpected Death Epilepsy (SUDEP). According to the World Health Organization (WHO) [1], nearly 50 million people of all ages around the world suffer from epilepsy. This condition significantly impacts the quality of life for these people.

Current treatment options for epilepsy are mostly found in drugs. These Anti-Epileptic Drugs (AED) work by controlling the seizures or by preventing them. AED works for 7 out of 10 people, making sure that they have no more severe seizures or stop seizures completely. However, for some people, these AED do not have the desired effect. This form of epilepsy is called Drug Resistant Epilepsy (DRE). For these people, other methods are possible to control the seizures. In general, AED is also known for the side effects that are rather unpleasant and can also alter a person's quality of life. Well-known side effects due to these drugs are tiredness, headaches, and nausea.

Other treatments are more invasive. One of these treatments, also known as resective surgery, is the removal of a part of the brain that is responsible for the seizures. These surgeries are often effective but have potentially severe implications. They could influence the way your brain works, with possible implications such as loss of vision, speech, memory, or movement. The other potential form of treatment is Vagus Nerve Stimulation (VNS). VNS works by sending small electrical pulses through the vagus nerve. A device is implanted under the chest with a connection to the vagus nerve in the neck. An example is shown in Figure 1.1. The device sends regular pulses to this nerve which as a result can lessen seizures and their severity. Al-



though VNS is a lot less risky than the other options listed before, Figure 1.1: Sentiva VNS, the price is often too high for many people and is therefore still not adapted from [2] considered too often in practice.

Less invasive methods for VNS have been researched to stimulate the vagus nerve in a similar fashion. Transcutaneous Vagus Nerve Stimulation (t-VNS) is a method of stimulating the vagus nerve without the need for invasive procedures. t-VNS can be applied to the Auricular Branch of the Vagus Nerve (ABVN) that is found in the ear. From here it travels through the vagus nerve to the Solitary Nucleus (NTS). t-VNS is used now more in a clinical setup and is as of now not approved for the treatment of epilepsy by the Food and Drug Administration (FDA). Besides that, there are not as many dedicated t-VNS devices that use a stimulation device intended for t-VNS solely. The goal of this project is to design a dedicated small and

power-efficient t-VNS device that would lay the basis for future development in the field of portable and wearable stimulators inside the ear.

### 1.2 State of the art for t-VNS

t-VNS is still in very early development, and although VNS has been FDA approved for the treatment of DRE, t-VNS still lacks the data to conclude that it has the same working as the original VNS [3]. An early proof of concept study had shown that seizure frequency was reduced in 5 out of 7 patients after 9 months of t-VNS therapy [4]. Another study had shown that after 8 weeks of treatment, t-VNS applied in the ABVN treatment resulted in a 42.6% average reduction in seizure frequency but it is not as potent as VNS [5]. Although VNS might be more potent, t-VNS is more cost-efficient and more accessible.

t-VNS has not only been researched for purposes of epilepsy treatment but could also be effective in the treatment of depression [6]. The devices that are used for t-VNS are mostly centered around a Transcutaneous Electrical Nerve Stimulation (TENS) device. A TENS is used not only for t-VNS but also for other forms of stimulation such as Peripheral Nerve Stimulation (PNS). The device that is most commonly used for t-VNS in the ABVN is the NEMOS Cerbomed [7]. However, this device has been discontinued from production due to undisclosed reasons. Other than the Cerbomed, a set-up with an ordinary TENS device and custom-made electrodes are often used in practice. However, none of these setups except for the NEMOS Cerbomed are specifically designed for t-VNS of the ABVN. This would mean that the current t-VNS setups cannot be translated to outside clinical practices.

Another shortcoming of most of the current devices for stimulation is the fact that they are all open-loop controlled [8]. This means that they stimulate at regular intervals throughout the day but not specifically when it is needed. Through the reading of biomarkers, the stimulation can be done in instances where it is needed the most.

### 1.3 Overview of the whole project

The stimulator is a subset of a bigger project where a brain-computer interface is made for an earpiece. Figure A.1 shows the different groups. The phantom group is responsible for making an *in vitro* resemblance of the electrical properties of human skin. The authentication group reads the Electroencephalogram (EEG) signals and based on these signals can perform authentication operations. Signal processing also reads EEG signals. But their goal is to detect and perhaps predict seizures in the onset of an epileptic attack. The stimulator operates based on the signals sent out by the signal processing group. This group sets the parameters and tells the stimulator device when it should or should not stimulate. This thesis will solely focus on the stimulation but it does take into account the different signals it will receive from the signal processing group.

### 1.4 Document lay-out

This thesis describes the design process and the implementation of the t-VNS device. Chapter 2 gives some important background information for the project that is required for the design of the device. Chapter 3 gives a list of requirements that were followed when considering the different implementation options for the stimulator. Chapter 4 gives an overview of the entire device and the different components it consists off. After that, the components from the overview are taken a closer look at and are described with the considerations and choices that was made in Chapter 5. Chapter 6 discusses the integrated circuit design for controlling the stimulator. In Chapter 7, the simulation results are discussed. The following chapters will describe the Printed Circuit Board (PCB) implementation and conclusions drawn from the results as well as recommendations for future work.

# Chapter 2 Background

The overall simplified overview of the system is seen in Figure A.2. To verify the correct working of the circuit, a testing set up is required. This testing set up will be consisting of the stimulation PCB attached to a phantom which mimicks the electrical properties of the human skin. However, in the beginning stages of the project, there is no actual human skin phantom to test with yet. Therefore, an electrical impedance model of the skin has to be made which is a simplified representation of the human skin. This model is also important for the design phase of the system, since the load gives an indication on how much voltage is needed to correctly stimulate the ear.

### 2.1 Impedance of the Skin

The skin is a layered system. In total, there are three promininent layers: epidermis, dermis and hyperdermis. However, the hyperdermis is not of significance since this is too deep in the skin and it does not contain any nerves to stimulate. Therefore, only the epidermis and dermis will be considered. The epidermis consists of multiple sublayers. The most relevant to the impedance model is the stratum corneum. This layer is predominantly responsible for the high impedance of the skin and should therefore be considered in the circuit [9]. It is also important to note that the stratum corneum does not have the same thickness everywhere on the skin. It is a lot thicker on the arm than it is on the forehead for instance.

The skin has been studied thoroughly and different representations have been made on how to model the electrical properties. The most commonly used model is the Montague Model, depicted in Figure A.3. This model captures the capacitive behaviour and the resistive behaviour of the epidermis. This model has been used in a variety of studies [10], [11].

There is also a non-linear model (seen in Figure A.4) which is based on the Montague model with the difference being that the resistivity of the epidermis is now a variable resistance [12], [13]. These models capture more of the non-linear behaviour of the skin due to the current intensity. Although these models might be more effective at capturing the non-linear behaviour, the models are too sophisticated and lack the relevant data to use it in this testing set up.

Although more models have been proposed for the modelling of the electrical behaviour of the skin [9], the Montague model is widely used to this day and is still shown to be sufficient enough to model most of the important characteristics. Therefore, the Montague impedance model is the model which will be used throughout the rest of this paper.

### 2.2 Electrodes

For stimulating the vagus nerve through the ear, electrodes can be placed in different positions. The goal is to optimally stimulate the vagus nerve. The location to place the electrodes needs to be found in which we

can both still use the earbud as a functional earphone and be able to still stimulate the vagus nerve properly. Comfort is an important factor for clinical practice.

The neural anatomy of the nerves inside the body is quite complex. The ear is connected to the NTS by the vagus nerve. The branch that is focused on is the auricular branch of the vagus nerve. This branch is the one that has the most afferent Vagus Nerve (VN) distribution which indicate that through this branch external stimuli can be exerted on the central nervous system.

The question on which side the stimulation should occur is also an important one. The left ear is chosen as the most common side of application because the right-side vagus has efferent fibers connected to the heart, which might result in cardiovascular side-effects [14]. Therefore stimulation will also occur through the left-ear only, since the safety is a very important requirement.

Research has indicated that stimulating the cymba conchae provides the most activiations in the NTS [15] and that the general concha area is very suitable for stimulation [16]. Stimulation of both the cymba and cavum conchae in the ear have been shown to have even more promising results [17]. The reason for this is that it is one of the main ABVN stimulation sites that is available in the ear and when stimulated this area produces the largest amount of activated ABVN fibers. What is also an important reason is the results that were found with the analysis of the different locations if the amount of activity that was found in the important areas connected to the vagus nerve, namely the Locus Coeruleus (LC) neurons and the NTS. This is an indication that stimulation at this spot resulted in the strongest vagal afferent pathway in the brainstem. It was found that the cymba conchae and cavum combination were capable of receiving a stronger intensity and, thus may offer an advantage over the inner tragus during long-term t-VNS treatment.

Not only is this position optimal for stimulation, it is also an ideal location to place the electrode considering it will be used in earphones. Earphones are often already placed against the cymba conchae and the addition of electrodes on the earphone can be placed in a manner that makes for a good integration. Therefore, the cymba conchae is the optimal place for the electrode to be placed. The size of the electrodes have to be in accordance with what would fit in the ear. The area that is desired is roughly 1 cm<sup>2</sup>. This would fit in the ear in the desired areas, namely one in the cavum conchae and one in the cymba conchae [17].

There are a variety of different types of electrodes: surface-electrodes (dry or wet), capacitive electrodes and needle electrodes. What distinguishes these electrodes is mostly in the the way they are connected through the skin. Surface electrodes are connected to the top of the skin with either no conductive gel or with conductive gel. Needle electrodes are electrodes that have very small needles that penetrate through the top-layer of the epidermis: the stratum corneum. In terms of impedances, electrodes can be considered as good conductors. This means that they have considerably low resistances. However, there are situations in which it does not just operate as a resistance.

### 2.3 The electrode-skin model for the testing ground

The connection between electrode and skin can be done in a variety of ways. As indicated, for each different type of electrodes there is a different model that is created. The model that will not be considered is the capacitive electrode. This is because capacitive electrodes for stimulating purposes require very high voltages that exceed levels that would be able to generate in a small earpiece. Therefore these forms of electrodes will not be used.

The wet surface electrode can be found in Figure A.5. The electrode in this model has a parallel capacitance. This originates from the double layer capacitance that occurs when an electrode comes into contact with an electrolyte gel. This causes a charge separation and therefore a capacitance. The electrolyte itself has a resistance and the impedance model is as was mentioned before in section 2.1.

The dry surface electrode can be found in Figure A.6. The reason that this electrode is purely modelled as a resistance is because the double layer capacitance is too small to consider. Therefore it is displayed as a

#### pure resistor.

Skin abrasion is not an option in the case of developing the model. The reason for this is that this would not be a suitable option since the stratum corneum regenerates regularly [18]. This would mean that every so often, the skin has to be abraded inside the ear. Skin abrasion is not comfortable and has to be repeated since the stratum corneum regenerates, which would make it a cumbersome activity.

Since skin abrasion is not an option, the stratum corneum will remain as an impedance in the model. Since the dry electrode does not improve the stratum corneum the impedance for this model will be too large to be considered [19]. The electrolyte for the wet electrode improves the impedance of the stratum corneum. This gives values that are better suited for the project [19]. The wet electrodes are suited for the project but they have problems. First and foremost is that gel has to be applied to the ear each time it is applied. The second is that they probably would not stick quite well when wearing them in the desired location, which makes it difficult to put them in the ear. For these reasons, wet and dry electrodes will not be considered for this project.

Then the only option that is left is that of a needle electrode. This electrode works by penetrating the top layer of the skin, which would cause it to skip the top layer of skin, the stratum corneum. The layers underneath the stratum corneum are far less resistive and is therefore a suitable final option. Since this can be seen as removing the stratum corneum (such as abrasion), values taken from [20] and using the same model for the needle electrode as for the dry electrode in Figure A.6, the used final model can be seen in Figure 2.1. Values can be found in Table A.1. Important to note is that there are different values noted that paper.



#### **NEEDLE FINAL IMPEDANCE MODEL**

Figure 2.1: Final impedance model accounting for 2 electrode connections

The reason that there is no  $R_{elec}$  is because relative to the other resistances, this is negligible. An electrode is considered a good conductor and is therefore not considered in this impedance model.

Although this is the correct model that should have been used, a different model was used to calculate the design parameters of the circuit elements. As such it will be used as the load characterization for the rest of the thesis. This old impedance model, however, did not differ a significant amount from the correct impedance model. This old model is shown in Figure A.7. The reason this model is wrong is because of

the fact the way the electrodes are modelled. The electrodes as they are depicted here resemble more of the impedance model as shown in Figure A.5. However, this only occurs when there is an electrolyte of significance in the layer underneath the electrode or when these values are known. During the process of designing this model, a confusion lead up to this mistake. Unfortunately, there was not enough time in the end to change this and therefore recommendations will be given for the future in chapter 10.

### 2.4 Wave Characteristics

Electrical nerve stimulation can be done with different stimulation wave forms. A large variety of research shows that square waves are commonly used for vagus nerve stimulation just like existing devices on the market. In addition, most research is available for the stimulation parameters of square waves [21], [22]. Furthermore, from the electronic design point of view a square wave is easier to generate in contrast to other wave forms like sine waves or triangular waves. As a result less components are required for generating the wave which could safe power as well. Therefore the stimulation waveform generated by the stimulator will be a biphasic square wave. The wave parameters of the square wave that will be used in this report are defined in Figure 2.2.



Figure 2.2: Definition of wave parameters

Research has shown that stimulation currents in the range of 1 mA until 10 mA are the most effective for t-VNS. However, Stimulation currents above 10 mA have to be avoided [17]. Stimulation frequencies in the range of 20-30 Hz are optimal for t-VNS, but stimulation frequencies of 50 Hz or higher have to be avoided since this can cause major and irreversible damage to the vagus nerve [23].

Regarding the pulse width of the stimulation wave form, the optimal value is still not known. However, pulse widths in the range of 100-500  $\mu s$  are proven to be effective for epileptic treatment. According to [24], a variety of research confirms that a pulse width around 200  $\mu s$  must work for epileptic treatment. Therefore a pulse width of 200  $\mu s$  is chosen for the stimulator.

Inter-phase gaps are often used in electrical stimulation devices to try to enhance neural stimulation efficiency by reducing the action potential. However, research has shown that introducing an inter-phase gap requires higher stimulation intensities compared to stimulation without an inter-phase gap to obtain the same results [25]. As a result, more power is consumed to achieve the same results. Since power efficiency is the main focus of this project, the inter-phase gap will not be present.

When stimulating human nerves, charge balancing is the most important safety factor. Applying a DC current to the skin for a longer period can cause pH change, electrode dissolution and tissue damage [26].

In order to avoid charge buildup, the stimulation waveform must be biphasic such that net zero charge appears on the electrode-skin interface. However, due to the capacitive electrode-skin interface, electric charge is stored during stimulation which can be solved by introducing passive charge balancing [27]. In other words, the load must be fully discharged before sending a new stimulation pulse to prevent charge buildup. Therefore the inter-pulse interval must be longer than 5 times the time constant of the electrode-skin interface such that the system is in steady state before simulating again to prevent charge buildup.

To get insight on the required voltage over the load for the discussed waveform, the following measurements have been done in simulations: Using the impedance model created in section 2.1 and a current source the voltage over the load can be measured. Due to the capacitive properties of the load, it is important that the current pulses sent through the load are similar (especially the pulse width) to the current pulses that were previously discussed. Measurements are done at the end of each pulse at the peak of the voltage. As described in section 2.1, the parameters for the load can be divided in three categories: The minimum load case, the general case and the maximum load case. The voltages at different drive currents for these cases can be found in table A.3. In conclusion, when taking worst case values for the load a maximum voltage of 65.8 V is required for stimulating with 8 mA.

# Chapter 3 Programme of Requirements

The requirements listed out in this chapter offer a basis as to how the stimulation should perform based on the research discussed in Chapter 2. The requirements are categorized in functional and non-functional. The functional requirements indicate how the stimulator must work where as the non-functional requirements indicate how the stimulator must perform. The functional and non-functional requirements are listed in sections 3.1 and 3.2, respectively. The requirements are explained in Section 3.3.

### 3.1 Functional Requirements

- 1. The system should be powered by a battery
- 2. The stimulator must be able to read the required stimulation current and frequency as input parameters at 3.3 V from a microcontroller
- 3. The stimulator must not inflict pain or cause an unpleasant feeling when used as indicated

### 3.2 Non-Functional Requirements

- 1. The stimulator must output a biphasic current-mode square wave
- 2. The current amplitude must not exceed 10 mA
- 3. Using the previously described impedance model of the skin as a load, the current amplitude must be adjustable in a range from 0 mA to 8 mA with a step size of 1 mA
- 4. The pulse frequency must be adjustable in a range from 10 Hz to 40 Hz with a step size of 5 Hz
- 5. The pulse width of the stimulation pulse must be 200  $\mu$ s
- 6. The inter pulse interval must be chosen such that the system is in steady state before simulating again
- 7. Active power consumption must be less than 1 W
- 8. Active and static power consumption must be minimized without compromising safety
- 9. The size of the stimulator must be minimized without compromising power consumption

### 3.3 Requirement Explanation

As mentioned earlier, the stimulation circuit will be placed in a wearable application. Therefore the power source has to be a compact battery.

Since the impedance of the ear is dependent of multiple factors and is highly subject to change, it is of high importance that the stimulator generates a constant current that is independent of the load in the range that is applicable for this in-ear application. As discussed in Chapter 2.1, in worst case a maximum of 65.8 V is required to stimulate with 8 mA. In other words, the stimulation current must remain constant in this voltage range.

Currents above 10 mA, or frequencies above 50 Hz must be avoided since adverse effects due to electrical stimulation are mainly caused by stimulating with higher currents or frequencies [17]. Therefore the stimulating device must shut down automatically when currents higher than 10 mA are measured. In addition, the device must send a fault signal indicating that it is not stimulating anymore.

For both research and clinical purposes, the device must have both adjustable stimulation current and frequency. The effectiveness for a certain stimulation current can differ per person. Therefore the stimulation current must be adjustable in a range from 0 mA until 8 mA with a step size of 1 mA. The frequency must be adjustable in a range from 10 Hz until 40 Hz with a stepsize of 5 Hz. So the device must able to read the required values for the stimulation current and frequency from a micro controller.

The main design focus of the stimulator is power efficiency. To be able to use an in-ear t-VNS device in practice, the device must be able to operate for a day or longer without replacing the battery. Therefore minimizing the power consumption is the most important factor when designing the stimulator. However, minimizing the power consumption must never affect the safety of the device.

# Chapter 4 Design Overview

The entire design overview of the stimulation device is depicted below. This overview shows all the components involved in the stimulator. These components will be briefly covered in this section and most of them will be elaborated further in other sections. The components that are not mentioned in future chapters are not designed by us and will therefore be covered in this chapter.



Figure 4.1: The design overview of the stimulator

#### Battery

The battery will provide the stimulator circuit and other all other components with power. Since the goal is to make this device portable and wearable, it is important to select a battery that has a high energy density, meaning that it takes up minimal area while still providing sufficient energy.

Lithium batteries are currently the best in terms of power and energy density on the market and is the most commonly used form of battery in modern technology [28]. One lithium polymer cell can deliver between 3.2 V - 4.2 V. The technology of lithium polymer is ideal for this project because it provides high energy density and is very lightweight. For wearable applications, this is ideal. They can also be very versatile in shape and size, which would make it very suitable for shaped earpieces and more

Since a high voltage of precisely 65.8 V (as can be seen in A.3) is required for the current source and a low voltage of 5 V is required for controlling the Field Programmable Gate Array (FPGA), a 3s Li-battery with a voltage between 9.6 V and 12.6 V will be used. This is suitable middle ground between the high and low voltage required for the operation of this device.

#### **Buck and Boost Converter**

Different components on the board operate at different voltages. 5 V is used to operate logic level components in the current source, H-bridge and other components in the ear piece. A high voltage (more than 65.8 V) is required to drive the load. Therefore, a buck and boost converter are both necessary to regulate the voltage from the battery (9.6 V - 12.6 V). The designs of these converters can be found in the Sections 5.4 and 5.5

#### **Impedance Model & Electrodes**

The stimulation will happen to the skin of a human being, specifically the ear. To accurately determine whether the design is working and the stimulation is according to the requirements a testing site has to be made. The electrodes that will be used for the stimulation also influence the impedance model. An in-depth explanation of this is already given in section 2.3

#### **Stimulation Source**

The stimulation source is responsible for safely sending an accurate current through the electrodes through the skin. To get an accurate current, a variety of measures have to be taken to make sure that the current through the skin does not depend on the impedance of the skin itself. A more in-depth explanation is given in section 5.1

#### **H-bridge**

In chapter 3, it was listed that the stimulation wave has to be biphasic and square. The H-bridge (also known as full-bridge) is a method of alternating the current through a load, thereby creating the biphasic behaviour that is desired. The design will be explained in section 5.3.

#### **Current Sense**

The current sense circuit is the part of the circuit that is responsible for measuring the current through the H-bridge. The reason of implementation of this component is efficiency (which will be discussed in sections 5.5 and C.5) and safety of the device. The design and working of the circuit will given in section 5.2.

#### **FPGA & Integrated Circuit**

The FPGA will be responsible for housing all the control logic. The FPGA will be used as testing device to make sure that not only the circuit but also the code that is going to be written is correct. The model that will be used for testing is the Digilent CMOD A7 [29]. Although an FPGA is used, it will only be used for testing. The goal is to house all the logic that will be written on the FPGA on a dedicated Application Specific Integrated Circuit (ASIC). This would remove the circuit board of the FPGA.

For writing the VHSIC Hardware Description Language (VHDL) code, Finite State Machine (FSM)'s are designed. This information can all be found in chapter 6

#### **Power Down**

The power down circuit is responsible for making sure that not only there is almost no static power dissipation but for also assuring that the FPGA starts up properly when it receives power. This is important as power efficiency is a big factor for this project as well as safety. This small component improves both of these factors and will be further explained in section 5.6.

# Chapter 5 Circuit Design

### 5.1 Stimulation Source

To send electrical pulses through the skin an electrical source has to be chosen. The stimulation source can be either a voltage source or a current source. Current amplitude is a very important parameter for stimulating the vagus nerve. The current controlled stimulation showed to have better controllability [30] [31] which is an important consideration since the human body is used as load. Also, current sources are in general load independent given the correct range of impedances that the load can achieve. Of course, if the load exceeds the maximum permissible amount and the voltage reaches a limit then the current will drop. Considering the fact that the stimulator should be load independent for the given range and given the controllability advantages of a current source, this will be the one that is used for the stimulator.

The current source has to adhere to the following characteristics:

- The current source has to be easy to control by an FPGA
- The current source should deliver a current with a margin of  $\pm~0.2~\text{mA}$
- The current source should have a very high output impedance making it independent of the load
- The current source must be adjustabel in a range from 0 mA to 8 mA with a step size of 1 mA.

#### 5.1.1 Current Source Implementations

The first option that was considered is the Howlard current pump. This is a very precise current source that is both able to sink and source current. A schematic can be seen in Figure 5.1.

![](_page_16_Figure_10.jpeg)

Figure 5.1: Howlard Current Pump

Figure 5.2: Wilson Current Mirror

The Howlard current pump can do both things for the use case of a vagus nerve stimulator, since it can generate a biphasic wave without the need for other components. However, there are some difficulties for

the usage of a howlard current pump for the purposes of the stimulator. First of all it requires a very high supply voltage to the amplifiers operate, which should be substantially higher than the voltage required over the load. Since the voltage that should be aimed for is around 65.8 V for maximum load, this would mean that a power supply voltage of roughly 70 V is required. Op amps with these specifications are extremely hard to find and consume a lot power. Besides that the resistors in the setup for the Howlard current pump required to be very precise since the feedback would otherwise be off which would result in an inaccurate gain. Due to these issues the Howlard Current Pump will not be used as the current source.

The other option that was found is a current mirror seen in Figure 5.2. A current mirror means that the desired current is "mirrored" to the other side of the circuit. The reference current pulls the transistors to a gate voltage such that the gate to source voltage is at the value for the reference current to flow through them. At the same time, it pulls the mirrored gate to the same voltage, forcing it to conduct the same reference current. As such, the current is mirrored. Important for the correct working of the mirror is that all transistors are in saturation and required gate to source voltage determining the current that flows through them. The Wilson current mirror is an adaptation of the original current mirror which improves the output impedance of the original current mirror.

The current mirror can be made with either a Metal-Oxide-Semiconductor Field-Effect Transistor (MOS-FET) or a Bipolar Junction transistor (BJT). To choose the right transistor for this purpose, the requirements are checked. The main requirement is power efficiency. As BJTs draw base current, power is dissipated for turning on the BJT. MOSFETs are more power efficient than BJTs and they are smaller in size as well. Although BJTs are faster, switching of the current mirror does not happen very quickly and is therefore not a priority for the design. Therefore, MOSFETs are chosen as the transistor technology

Multiple versions of current mirrors can be made. Besides the Wilson current mirror, the cascode current mirror is also an option. However, it was stated in a paper that a Wilson current mirror had the highest bandwidth out of the two mirrors with negligible differences in other areas [32]. Since the cascode current mirror had similar performance in terms of output impedance and error accuracy, this could have also been used. Both adaptations are acceptable in terms of performances.

To further improve the Wilson current mirror, it can be modified to provide multiple outputs (BJT wilson current mirror with multiple outputs given in [33]). This mirror can be seen in Figure 5.3. The reason for adding multiple outputs is because the reference current can then be smaller. The reference current also consumes power but that power is considered wasted as it is not directed towards the load. The efficiency of the Wilson mirror in Figure 5.2 is roughly 50%, since the same current flows through both reference and output both originating from the Vdd. With the improved version in Figure 5.3, this efficiency rises to roughly 83.3%. This is a significant improvement over the original amount. However, this brings up the question why not add in more than 5. This is due to the constraints of size and efficiency. Adding additional transistors would significantly limit the amount of space on the board. This would also not increase the amount of efficiency that significantly. Doubling the amount of outputs would give an increase of 6.7%.

In a current mirror, the reference current determines the current that is mirrored to the other parts of the mirror. It is important that this reference current can be controlled. This would require a variable voltage source to be used. The current range that is given in chapter 3 mentioned that stimulation from 0 to 8 mA should be controlled in steps of 1 mA. The solution for this problem is given below.

The Digital to Analog Converter (DAC) takes a serial bit value as its input from which the reference voltage can be derived. This gives the corresponding output voltage which maps to the desired current with the following formula:

$$I_{ref} = \frac{V_{in}}{R_1} \tag{5.1}$$

To calculate the resistance value  $R_1$ , the equation above is used. Given the maximum value of the DAC (which is 1.24V) and the maximum reference current, a value of 775  $\Omega$  is retrieved. To make sure that the resistance always falls in the range of the DAC, a resistance of 681  $\Omega$  is used as the reference resistance. The complete current source is displayed in Figure 5.3 with the reference current source

![](_page_18_Figure_1.jpeg)

Figure 5.3: Complete current source with current mirror and reference current

#### 5.1.2 Results

The power efficiency is almost always exactly the 83.3% that was mentioned before. Figure 5.4 shows that the current is correctly mirrored. It can be seen that at the timestamp shown, the current is almost exactly 5 times the reference current. This means that it is working as expected. Noteworthy is the delay between the reference current going to its appropriate value and then the current mirror going to the appropriate value. This delay is most likely an issue due to the fact that the gates of the mirrors need to be pulled to the

![](_page_18_Figure_5.jpeg)

Figure 5.4: Reference current of 0.2 mA mirrored and multiplied 5 times

level needed to conduct the reference current. This takes time since there are a lot of gates that need to be charged. In the final design this will not be much of an issue since the current mirror has a startup phase before going through the load. This means that the current is at the correct level as soon as stimulation is supposed to occur. This will be further explained in section C.2.

Table B.2 in the appendix shows the currents and the error that are measured through the old load model as indicated in section 2.3 with the maximum voltage of 70 V as supply as indicated in section 5.5. The shift from positive undershoot to negative undershoot is one that is difficult to explain. A possible explanation for this might be due to the fact that when there is more current flowing through the transistors, the gates of the transistors need higher voltages for the currents to match up proportionally. This draws current away from the other gates, which could result in the drop in the current. These deviations can be accounted for by changing the value through the DAC, which gives a slightly higher or lower reference voltage to get a higher or lower reference current. The table does indicate that the current source in Figure 5.3 is working correctly.

In terms of power dissipation, Table B.1 shows the results measured for different currents with a constant voltage source of 75 V. Noteworthy is the fact that the power efficiency decreases a lot when the stimulation current is low. This is mostly due to the fact that there is a lot of dissipation in the mirror when there is stimulation from a constant voltage source. When the maximum current is achieved, the power efficiency is at a very reasonable value of roughly 75%. With this comes a power drawn from the source of 351.73 mW.

### 5.2 Current Sense

The current sense accurately measures the current through the H-bridge and current mirror which corresponds to the current through the load during a stimulation pulse. With this information, the FPGA can detect undercurrent and overcurrent which is necessary for proper functionality of the circuit and to satisfy the mandatory safety requirements.

The FPGA which is used in this design has a built in Analog to Digital Converter (ADC) which translates voltages between 0 V and 1 V to a 12-bit binary number. It will be used to create binary information from the current. The ADC expects an input of 0 V to 1 V which means that the current range that is of interest should be mapped to this voltage range. In this case we will consider a current range of 0 mA to 10 mA since 0 mA to 8 mA is the active current range in which stimulation takes place and 10 mA is the maximum allowed current according to the requirements.

#### 5.2.1 Current Sense implementations

Mapping current to voltage can be done with a sensing resistor, which is a small resistor in the series path of the current that has to be measured to induce a small voltage over it. Three different implementations were considered. They can be found in figure 5.5.

![](_page_19_Figure_8.jpeg)

Figure 5.5: The three considered current sense implementations

The first and the simplest implementation chooses Rs in such a way that 10 mA of current through Rs will create 1 V over Rs (Rs=V/I). This implementation is only accurate if the FPGA's ADC has a high input impedance and thus draws an insignificant amount of current. However, this is not the case, the on-board ADC functions by charging and discharging a capacitor which consumes current. How much is not stated in the datasheet [29]. Measurements will be less accurate when not adding a buffer. Placing the sensing resistor under the H-bridge also increases the ground potential seen by the H-bridge. In this case by 1 V in case of a 10 mA current relative to the real ground. Depending on the transistors used in the H-bridge, this could influence performance. Excluding the power consumption of the ADC, the circuit would dissipate at most 8 mW.

The second implementation seperates the output voltage and the voltage over Rs by means of an operational amplifier. Because of the high input impedance on the input of the amplifier, current over Rs is mapped to a voltage in a very accurate way since practically all current flows through Rs. Furthermore, the operational amplifier is able to amplify the voltage over Rs which allows Rs to be smaller to get the same voltage range of 0 V to 1 V on the output of the amplifier which reduces the power dissipation in Rs. Decreasing Rs reduces the rise of the ground potential seen by the H-bridge which allows for faster switching times and an increase of the rise and fall time. This implementation trades Rs size and simplicity for a more accurate current measurement and a decreased voltage rise of the ground seen by the H-bridge.

In the third implementation option, the sensing resistor is placed above the H-bridge eliminating the voltage rise of the ground seen by the H-bridge which means that the gate to source voltages on the MOSFETs are always exactly the same and as high as possible. However, this method comes with a significant trade off which is introduced by the differential amplifier. This amplifier needs to handle the voltage levels at Rs relative to real ground which could be as high as 70 V. These high voltage amplifiers are very inefficient, often not rail to rail, very large and they are due to the chip shortage often not available. Therefore this is not an option for this project.

To maximize power efficiency (to satisfy the requirements in chapter 3.3) and minimize the rise of the ground potential of the H-bridge, implementation 2 will be chosen. Rs will be minimized and the voltage gain of the amplifier will compensate for the decrease in voltage over Rs. A 10x gain was chosen since from a simulation it appeared that the difference in circuit characteristics of the H-bridge was negligible at 100 mV ground potential compared to 0V ground potential. The component values for this implementation can be found in the appendix in table B.5.

#### 5.2.2 Simulation Results

The designed circuit was modelled in LTspice (without a model for the FPGA) for currents between 1 mA and 10 mA. For each measurement, the output voltage, the current supplied to the operational amplifier and the voltage over Rs were measured. With these measurements, the total power consumption has been calculated. The results can be found in table B.6.

As can be seen from the last column, the maximum power dissipation in the active stimulation range occurs at 8 mA drive current and is equal to 1.8 mW which means that this implementation surpasses the first implementation when considering power consumption. Important to note is that the bias voltage of the amplifier is clearly visible in the output voltage which is read by the FPGA. The output voltage is in most cases 2 mV and in some cases 1.9 mV lower than calculated. This deviation can be accounted for in the FPGA.

### 5.3 H-bridge

An H-bridge is a simple and power-efficient way of transforming a Direct Current (DC) into a square wave through the load. Therefore, an H-bridge circuit will be used for transforming the DC current generated by the current source into a biphasic square wave through the load. However, due to the high voltages regulated by the current source in the order of 66 V, the high-side MOSFETs require high gate voltages to operate. This is a problem since the control signals generated by the FPGA will operate at only 3.3 V. There are multiple solutions to tackle this problem.

#### 5.3.1 H-Bridge Implementations

Two possible implementations of high-side switching circuits are shown in Figure 5.6 and 5.7.

![](_page_21_Figure_5.jpeg)

Figure 5.6: High-side switching circuit Figure 5.7: High-side switching circuit with bootwith a P-channel MOSFET strap

In Figure 5.6 a high-side switching circuit based on a P-channel MOSFET is shown. Here,  $V_{gs}$  is either 0 V or the negative supply voltage generated by the current source. However, power efficiency is a problem when high voltages in the range of 66 V or even more are regulated by the current source. Simulations showed that R1 must not exceed 10 k $\Omega$  to achieve acceptable switching times. As a result, as soon as the gate of Q2 is pulled to Ground (GND), a current of 6.6 mA is leaking straight to GND at 66 V. The power loss due to the current leakage to the GND with R1 = 10 k $\Omega$  is calculated below. The power loss due to Q1 is neglected.

$$P_{loss} = \frac{V_{source}^2}{R_1} = \frac{66^2}{10 \cdot 10^3} = 435.6 \ mW \tag{5.2}$$

Another possible solution is using a bootstrap circuit to drive the gate of an N-channel MOSFET as shown in Figure 5.7. This circuit operates as a step-up charge pump which adds the supplied voltage over C1 to the applied voltage at the drain of Q1 resulting in the gate to source voltage being equal to the voltage over C1. Simulations showed that R1 should have a value of at most 10 k $\Omega$  to achieve fast switching times. This implementation is way more power-efficient since the voltage drop over R1 is approximately 4.3 V when shorted to GND (assuming a voltage drop of 0.7 V over D1), resulting in a power loss due to the current leakage to the GND through R1 of only 2.73 mW when taking the base current of Q2 into account as well.

Finally, optocouplers were considered for high side switching. However, optocouplers typically require more than 1 mA forward current which will leak straight to GND. At 3.3 V the power loss for this application will be 3.3 mW or more, which is already higher than the power loss of the bootstrap application due to current leakage to the GND. In addition, optocouplers with low forward current were mostly not available at this moment.

In conclusion, the bootstrap circuit is the most power efficient way for driving the high-side MOSFETs. The final implementation of the h-bridge with bootstrap circuits is shown in Figure 5.8.

![](_page_22_Figure_2.jpeg)

Figure 5.8: H-bridge with bootstrap circuit

As can be seen in Figure 5.8, the bootstrap circuits are connected to a 5 V supply coming from the Power Down Circuit, which will be discussed in Section 5.6. The MOSFETs are all separately controlled by the FPGA to be able to control switching timing independently. The switching timing will be discussed extensively in Section C.2. Furthermore, the gates of the high-side MOSFETs are connected to a pull-down resistor and the gates of the low-side MOSFETs are connected to a pull-up resistors are connected to the 5 V buck converter which is always turned on in contrast to the 5 V supply from the power down circuit. As a result, when no simulation is performed the high-side MOSFETs will act as an open circuit and the low-side MOSFETs act as a short circuit. This way the capacitive load is able to discharge through either D4 and Q3, or D3 and Q4 to ensure charge balance in the skin.

#### 5.3.2 H-Bridge Simulations

The h-bridge as shown in Figure 5.8 is simulated together with the current source as discussed in Section 5.1, and the impedance model with worst case values as discussed in Section 2.3. The final selected components are summarized in Table B.4. The current source is configured such that it produces a current of 8 mA. The simulation results are shown in Figure B.1. As can be seen in the figure, the h-bridge approximately generates a square wave through the load. However, a current peak is present at 0.4 ms which can be explained by the discharging capacitive load. From 0.2 ms until 0.4 ms the capacitive load is charged resulting in a current peak due to discharging when the polarity is switched. Furthermore, at 0.6 ms the discharging effect of the load can be observed as well. Therefore the inter-pulse interval is required to ensure charge balance.

The power loss of the h-bridge can be obtained by comparing the power generated by the sources and the power dissipated by the load. The average power generated by the sources and the power dissipated by the load were found by measuring the average voltages and currents during continuously stimulation. The cumulative average power generated by the sources is 235.8 mW while the power dissipated by the load is 231.2 mW resulting in a power efficiency of 98.05%

#### 5.4 Buck Converter

The FPGA, the H-bridge, the Stimulation source and other potential circuitry that will be present in the ear-piece (which is out of scope for this thesis) all require a stable 5 V supply voltage for proper operation. It is therefore essential to regulate the voltage of the battery. Besides the requirements discussed in chapter 3, this DC/DC converter has the following requirements:

- Input voltage range of 9.6 V 12.6 V
- Output voltage regulated at 5 V with a maximum deviation of 100 mV
- Maximum output current should be 3 A

#### 5.4.1 Buck Converter Implementations

Since battery voltage is in all cases higher than the output voltage, a step down voltage regulator will be used. Two different implementations have been considered in the design.

The easiest way to step down voltage is by using a linear regulator. A Linear Dropout regulator (LDO), is an example of such a linear voltage regulator. They are small in size, very accurate, available as an Integrated Circuit (IC) and they are often used when the supply voltage is close to the regulation voltage. This is usually the case since the power consumption of the LDO increases linearly with the difference between the supply and output voltage. According to [34], the power dissipation can be described by the following formula where  $I_o$  and  $V_o$  are the output current and voltage,  $V_i$  is the input voltage and  $I_q$  is the quiescent current of the regulator.

$$\eta_{LDO} = \frac{I_o V_o}{(I_o + I_a) V_i} \cdot 100\%$$
(5.3)

From this equation can be derived that in the best case, with a battery voltage of 9.6 V and no quiescent current, the efficiency is 52.1% and only decreases when the battery voltage becomes higher.

Another option is to use a switch mode step down converter such as a buck converter. Buck converters consist out of an electrical switch, a diode, an inductor and a capacitor. They are relatively large in size but they can, depending on the load conditions, reach an efficiency of over 90%. Since efficiency is the main focus of this thesis, the buck converter implementation is chosen. The LM2576 regulator IC is used to minimise the size of the buck converter and reduce the designs complexity. Due to the chip shortage, higher efficiency buck converter topology IC's were not available. The efficiency at 12 V input and 3 A load current is 75% according to its datasheet [35]. The typical 5 V application circuit schematic from the datasheet can be found in figure 5.9 and the corresponding component values can be found in table B.7

![](_page_23_Figure_12.jpeg)

Figure 5.9: LM2576 5 V regulator schematic

#### 5.5 Boost Converter

The boost converter will supply the voltage that is required by the load and current source to be driven at the specified drive current. From the results in table A.3 it becomes clear that the voltage range over the load at the desired stimulation currents is very large and ranges from 2.16 V and 65.8 V. A practical, non-ideal current source such as the Wilson current mirror that was discussed in section 5.1, 'absorbs' the difference between the supply voltage and the voltage over the load at the required drive current which means that there is power dissipation in the current mirror which is linearly proportional to the difference between the supply voltage over the load. To minimize power dissipation and improve the efficiency of the design, a digitally adjustable voltage regulator which minimizes the voltage over the current mirror is implemented in the design.

The following requirements are relevant for the design of the boost converter:

- Input voltage range of 9.6 V 12.6 V
- Output current of 0 mA to 8 mA
- Design for the complete load voltage range but optimize for the average case.

#### 5.5.1 Boost Converter Implementations

Considering the battery has a voltage between 9.6 V and 12.6 V and the voltage range of the load is between 2.16 V and 70 V one could say that it is straightforward to use a buckboost converter which is able to step up and step down the voltage. This is not necessarily the case. The converter should be designed to provide a high enough voltage to provide the load over the entire impedance range to ensure the stimulation current range but for better power characteristics, the general case should be optimized. Having the option to step down power only optimizes the general case with a drive current of 1 mA. However, this is not directly a reason to not use buckboost. The downside of using a buckboost converter is it's duty cycle which is always larger than the duty cycle of a boost converter. Large duty cycles cause larger power losses. Furthermore, buckboost converters require an electrical switch which is not connected to ground. This complicates the circuit, takes up space and consumes power as well. For these reasons, a boost converter is chosen.

Boost converter IC's similar to the buck converter IC discussed in section 5.4 are unfortunately not available due to the chip shortage. There are very few IC's available that can handle 70 V output voltage and many of them were out of stock or could not deliver the desired voltage range. Therefore, the choice was made to generate the Pulse-Width Modulation (PWM) signal for the switch on the FPGA. This also makes it easy to actively change the duty cycle to control the output voltages according to the load. The circuit schematic of the boost converter can be found in figure 5.10

![](_page_24_Figure_10.jpeg)

Figure 5.10: General Boost converter circuit schematic

#### 5.5.2 Simulation results

Simulations have been done in LTspice. A non ideal current source has been connected as the load to simulate the load the boost converter should be able to handle in the real application. The simulations are done with LTspice models of real components. A list of those can be found in table B.8 and their value calculations can be found in the appendix in section B.5.1. The values for the capacitor and inductor are larger than calculated to create headroom for possible inaccuracy in the components or in the formula's. To determine the worst case performance of the boost converter, simulations were done for every possible duty cycle and drive current with an input voltage of 9.6 V. The following parameters were measured: output voltage (table B.10), current through the zener diode (table B.11) and input current (table B.14). From these simulations, the efficiency can be calculated which can be found in table B.15.

These same simulations have been done for the maximum input voltage of 12.6V but only for the the high duty cycle cases to get insight in excessive voltages and currents. These results can be found in the tables B.16, B.17, B.18, B.19 and B.20.

The first thing that should be noted is the steady state current in the zener diode. When currents become too high for an extended amount of time, the power rating of the zener diode may be exceeded. The power rating of the used diode is 1 W which relates to 13.3 mA at the zener voltage of 75 V. This value for the current is only exceeded at 85% duty cycle at 12.6 V input. Running the boost converter at this duty cycle will damage the circuit. A possible solution for this problem would be to allow a higher voltage output by replacing the zener diode by one with a higher zener voltage. However, this is not possible due to high voltage limitations in the current mirror and H-bridge. It was decided to change the maximum duty cycle to 80%. As a result, with a worst case load and a 9.6 V battery, a drive current of 8 mA cannot be achieved. This is acceptable since the entire general case drive current range can be achieved together with 1 mA - 7 mA drive current with the worst case load conditions. Furthermore, the FPGA has a safeguard against undercurrent and will shut off when undercurrent is detected during multiple consecutive pulses.

Furthermore, the voltage peaks and the average current peaks through the zener diode should be considered. As can be seen in B.17, the average current during the initial peak is currently 60 mA. This exceeds the power rating of the zener diode. A solution for this is a soft start sequence in which the duty cycle is slowly stepped up to prevent voltage overshoot and thus decrease the current peak through the zener diode. Such a Soft start sequence with a maximum duty cycle of 80% and an input voltage of 12.6 V can be found in the appendix in Figure B.2. The average peak current with this sequence is 12 mA which is within specifications of the selected zener diode.

To further decrease power consumption of the design, the maximum duty cycle at which the soft start sequence ends depends on the drive current. Using the worst case voltage output results of the boost converter, which can be found in Table B.10, to look up the minimum duty cycle required to achieved the worst case load voltages which were presented earlier in Table A.3, a table for  $D_{max}$  as a function of drive current has been constructed. This table can be found in the Appendix C.2.

The power consumption and efficiencies for the general case are evaluated as well. A summary of the calculated efficiencies for the general case boost converter output power can be found in Table B.9. The efficiency of the output power compared to the input power of the boost converter can be found in the fourth column, the efficiency of the load power compared to the input power can be found in the fifth column.

Lastly, the no load power consumption was measured. Current measurements were done in LTspice with several duty cycles and input voltages. The results can be found in Table B.21. It can be observed that for the absolute largest range of the duty cycle and input voltage, there is no load power dissipation. This is due to the fact that the voltage does not rise fast enough to exceed the 75 V zener voltage. Furthermore the power dissipation is within specs of the Zener diode dissipation of 1 W.

### 5.6 Power Down Circuit

The power down circuit is responsible for making sure that when there is no power needed for the stimulator, no unnecessary power will flow. The reason this component is put in there is because of the power efficiency. Disconnecting the power source from all the components in the circuit is a very safe method of making sure no static dissipation occurs. As can be seen in Figure 4.1, when the battery is disconnected from the circuit, none of the components will have any supply connected to them. This is shown in figure 5.11.

![](_page_26_Figure_3.jpeg)

Figure 5.11: Diagram of the connection of the power down circuit

The most important factor that needs to be taken into account is the FPGA losing power. When the FPGA loses power, the states don't automatically return to the reset state. Turning on the power suddenly can cause the circuit to exhibit behaviour that is undesired. Therefore, a reset has to be implemented such that the circuit goes into a known state from which it can go to other states.

Although a reset circuit could be designed from scratch, it is more space and power efficient to use a power on reset timer that is readily available on a chip. The chip that is used for that purpose is the MAX810 by OnSemi [36]. The chip works by getting the supply voltage of the FPGA as an input which it compares to a given reference voltage (in this case 4.5V). When the supply voltage exceeds this voltage, a rail to rail input amplifier initiates a timeout counter. This counter makes sure that the reset is high on the FPGA for a given time, fulfilling the role of having a reset when the circuit is just powered on.

The amount of consumed power is difficult to estimate, since it would require knowledge about the exact static power dissipation of the other components in the circuit. The power down circuit draws power when there is the need for switching from the off to on state or vice versa. A CMOS is known for its high input impedance which results in practically zero gate current. Therefore, the power dissipation in the power down circuit only uses power for switching which would be a lot smaller than continuously connecting the battery to the components.

## Chapter 6

## **Integrated Circuit**

![](_page_27_Figure_2.jpeg)

Figure 6.1: Submodule layout of the integrated circuit

In this chapter, the IC's functionality will be discussed. The entire overview of the to be designed integrated circuit can be found in Figure 6.1. This shows how the different sub-modules are interconnected with one another. The dotted lines are signals that are sent or received from outside components. Note that the microcontroller listed in this figure equals the signals received from the signal processing group as mentioned in section 1.3. The ADC is located on the FPGA and the microcontroller is a stand-alone circuit. The arrows going out of the box but solid are the lines that are sent to the actual stimulator circuit.

The IC has control over the amplitude of the stimulation current, all four branches of the H-bridge and the output voltage of the boost converter. In addition, it reads the current through the H-bridge and it is able to read commands from the microcontroller controlled by the signal processing group. It is also able to detect improper functionality by interpreting the measured current and it will send a fault signal to the microcontroller if this is the case. During the design, special attention was give to guarantee safety and power efficiency.

### 6.1 Safety

The microcontroller has no direct control over the control signals going to the circuit. Instead, they are decoded into safe parameters. Implementing this in hardware guarantees safe operation of the electronics in the stimulator and makes the design robust against potential hacks of the microcontroller.

Furthermore, the main control sub-module will disable stimulation automatically when the measured current deviates from the expected current range. This makes the system robust against circuit failure in the current source and boost converter. It also prevents unsafe operation when the device is misused by a patient or when the device operates in other load circumstances than those that were accounted for in the design. For example when the electrodes are connected to a load which is too large.

Lastly, During the inter-pulse interval, the H-bridge control will configure the H-bridge in such a way that the load is completely disconnected from the boost converter but stays connected with ground. This way, any charge left in the skin can safely discharge via ground.

### 6.2 Efficiency

As discussed in section 5.5, for higher efficiency the boost converters output voltage should be as close as possible to the voltage over the load. The control logic is implemented in the Boost control sub-module. The algorithm works as follows: Initially, the duty cycle is stepped up gradually until it reaches the maximum allowed duty cycle indicated by the decoder (which depends on the drive current). Then, the duty cycle is slowly stepped down after every stimulation pulse until undercurrent is detected. When undercurrent is detected, the duty cycle control sets its value to last duty cycle when no undercurrent was detected. The duty cycle optimization is done and the value for the duty cycle is maintained. Another undercurrent event will reset the algorithm. Furthermore, to save additional power, the current source will be disabled during the inter-pulse interval.

# Chapter 7

## Discussion of the Results

### 7.1 Simulated Results

Estimates of the total average power dissipation of the entire circuit have been given in previous chapters. It is important that the power consumption requirement as indicated in chapter 3 of 1 W is adhered. Therefore the different components maximum power dissipation will be checked. Table B.9. For the current of 8mA, the boost converter uses a power of 439.7 mW. In reality, however, if the loads draws 8 mA then the current drawn from the boost converter is roughly 9.6 mA since the reference current also needs to be generated. So using linear extrapolation on the 439.7 mW for 8 mA, the power for 9.6 mA is roughly 527.64 mW. This is the power supplied by the battery of the circuit during stimulation. The average power over the load during maximum current amplitude stimulation is according to Table B.1 around 212.83 mW. Also using the efficiency of 98.05% calculated in section 5.3, the total power amounts to 538.41 mW The requirement that the total power during active stimulation is under 1 W has been validated since the power used is around the 538.41 mW for maximum stimulation.

This does not account for static power dissipation in the components that are not included in this analysis. The power consumption of the current sense is minimal as indicated in section 5.2. The power the current sense uses is in the orders of 1.8 mW for maximum drive current. The power the buck converter consumes is highly dependent on how much the FPGA drains it but also the op amp that is used for the reference circuit. The op amp itself draws about 50  $\mu$ *A* of quiescent current with a voltage of 5 V. This amounts to roughly 0.25 mW of turn on power for the op amp, which is negligible relative to the main circuit. This analysis so far only looks at the period when there is current flowing, not the entire stimulation period. When there is no active stimulation (meaning that the circuit is in the inter-pulse interval) the boost converter still supplies power to keep the voltage high. This can be found in Table B.12. The worst case average power dissipation is roughly 811 mW during the inter-pulse time . Sine the inter-pulse time is longer than the pulse times, this is also the average power during stimulation for the worst case. This is significantly higher than required for the future purposes but still remains under the 1 W requirement that was given before. For a general case, the no load average power dissipation is around the 160 mW for 9.6V with 75% duty cycle.

Unfortunately, due to issues with the simulating software there is no simulation of the entire circuit. When analyzing the behaviour of the simulations of the total circuit, the op-amp of the reference current source stopped functioning and started clipping. Also connecting a large current mirror to the H-bridge caused current spikes and drops that were not in line with the expectations. It was seemingly like the H-bridge was acting like a load seen from the current mirror which is not how an H-bridge should act.

### 7.2 Prototype Results

The results of the prototype were not according to the expectations. Mostly due to the short span of time to test and implement the prototype there was no option to go in-depth into each separate component to

#### 7.2 Prototype Results

debug it.

The most thoroughly tested component was the H-bridge which had the most unexpected results arise. The setup was as follows:

- 5 V supply for the bootstrap network
- 20 V supply for the load
- 124 Hz signal for the H-bridge gates
- 18 k $\Omega$  functioning as the load

On power up, the system did not behave as expected. The circuit consumed more current from the 5 V rail than previously expected. Performing several measurements confirmed that there was a short circuit caused by a PCB fabrication error at a MOSFET from the power down circuit. This component was desoldered which solved the short circuit.

However, after fixing this problem, there was current flowing through the H-bridge that was not going through the load which should not be possible in the test setup that was used. Testing indicated that when all transistors were open circuited, there was no current flow. However, when the high-side transistors were closed, there was current flow which in theory should not happen. Therefore, the lower side of the H-bridge was identified to be the issue. The bottom-left part of the H-bridge always gave 0 V meaning that this was the side that was connected to ground and caused the short circuit. So it was either the transistor or the diode. Having both diodes from the H-bridge removed did not fix the problem. A possible cause is a nonfunctional MOSFET on the left side of the H-bridge. Unfortunately, this theory could not be confirmed due to time constraints.

VHDL was written for the FPGA to implement the integrated circuit as indicated in chapter 6. Although the VHDL worked correctly in simulations, the synthesis of the VHDL on the FPGA was not as successful. First issue was that the ADC was not easily connected and mapped to the VHDL which gave as a result that the entire circuit could not be tested since so many components depend on the current sense, directly or indirectly. A separate testbench for the boost converter was written to test the soft start functionality. However, the FPGA did not generate the expected PWM frequency for the boost converter that was confirmed by simulations. The frequency was around 4 MHz instead of 600 kHz. This caused the inductor to heat up and break. Also what did not help is that when the bitstream was written to the FPGA through JTAG, it did not flash it properly and the programming was gone.

The current mirror was also tested without the use of the DAC (due to the FPGA not working as expected). A voltage reference was manually generated by adding a voltage source. Results indicated that the op amp did get power, however, it did not supply any voltage to the BJT, causing no reference current to be generated. Since the reference current is the main factor for the functioning of the current mirror, there was no indication whether the mirror actually functioned. The expected culprit is the op amp due to perhaps poor connection or overheating during fabrication, since an identical op amp was used to the ones in simulations.

# Chapter 8

## Prototype implementation

The discussed design of which its functionality is proven by simulation will be implemented on a PCB. To make the design functional and power efficient, all available time until the thesis deadline for the project was used for design and simulation. According to the planning, the time between the thesis deadline (June 13th 2022) and the defense (June 20th 2022) will be used to construct the prototype PCB, write VHDL and perform tests so that results can be presented during the thesis defense (June 20th 2022). In this chapter, the PCB design will be briefly discussed.

### 8.1 PCB Design

Due to the complex design, it was chosen to design a 4 layer PCB. The top and bottom layers will be used for traces between the components, while the middle two layers will be used as a ground plane and a 5 V plane to power all 5 V components. In addition, all free space left in the top and bottom layers of the PCB will be filled with ground planes to ensure minimal impedance from components to ground.

The circuits with their corresponding components can be subdivided in three categories: digital (FPGA and DAC), stimulator (Current source, H-bridge Current sense) and power (Power down, Boost converter and Buck converter) To avoid interference between different components, special care is taken to separate high current and EMF producing components from low voltage and digital signals. This can be seen in the separation of ground planes. The three categorized circuits are all connected to a different part of the ground plane and only connect to each other close to the ground of the supply voltage (battery) connection. Furthermore, the inductors from the boost and buck converter have been placed on the backside of the PCB to separate them from the digital and low voltage components to avoid interference from EMF waves.

The PCB has been designed in such a way that all different sub-circuits can be disconnected and tested separately through pin headers. In addition, the MOSFETs controlled by the integrated circuit can be disconnected from the FPGA and connected to a micro controller in case the integrated circuit fails. Furthermore, test pads were added at all relevant nodes for testing purposes. As a result, the PCB is much larger than in a design where no testing capabilities are required. In practice the FPGA would be replaced by a much smaller integrated circuit as well. Furthermore, due to the chip shortage, multiple IC's were used with P-channel and N-channel MOSFETs combined. In multiple cases only one of the two was used. In a later final design this could be prevented, or multiple used P-channel and N-channel MOSFETs could be combined on one integrated circuit to save space.

# Chapter 9 Conclusion

For this TU Delft bachelor graduation project an electrical circuit for t-VNS has been designed and implemented on a PCB. Test results of the PCB were not included since testing was scheduled in the period between the bachelor thesis deadline and the thesis defence. The electrical stimulation circuit was designed around an impedance model of the skin. A large variety of research has proposed different impedance models. However, the stimulator circuit was designed around the Montague impedance model taking the worst case impedance values into account. Simulations showed that the stimulator is able to approximately generate a current-mode biphasic square wave through the load. However, since passive charge balancing has been applied the generated square wave was not perfect due to capacitive behaviour of the load.

The stimulator is able to accurately stimulate with currents until 8 mA when taking worst case values for the impedance of human skin and general case values for the battery. However, Due to the limitation in duty cycle of 80% of the boost converter, a drive current of 8 mA could not be achieved with a worst case load and a 9.6 V battery. However, this is acceptable since the requirement of 8 mA can be achieved in the most cases and safety measures are in place to guarantee safe stimulation. However, all other requirements regarding the wave parameters such as the pulse width, stimulation frequency and thus stimulation current in general case have been met.

Overall, the individual components are performing as intended. The current mirror is correctly multiplying the reference current and giving the desired magnitude for the current. Improvements can be made in terms of the pace at which it goes to the correct current. The reference current source is very stable and behaves according to expectations as shown in Figure 5.4. The H-bridge is successful in alternating the current as shown in Figure B.1. The power down circuit unfortunately did not have a spice model to simulate with and was therefore not able to be fully verified. However, taken the assumption that it is from a verified company, this should work as intended.

Furthermore, an integrated circuit has been designed for controlling the stimulating circuit based on required input parameters. The integrated circuit has not yet been implemented. However, this will be implemented on an FPGA for the thesis defence. The integrated circuit has, among other things, built in safety features such as over-current protection due to closed loop feedback in current sense.

Together, these functioning submodules make for a portable, power efficient and very safe transcutaneous vagus nerve stimulation device. This device is suitable to lay the basis for future development in the field of portable and wearable stimulators inside the ear.

# Chapter 10 Recommendations & Future Work

The designed stimulator circuit is very much designed around the impedance of the skin. The research done for this device with regards to the impedance was numerous but detailed information was not found. Assumptions were made with regards to the impedance. Aspects such as different thicknesses of different layers in various parts of the body were not taken into account. In the future this model has to be made more accurate and has to be made to resemble the skin in the ear. When this model is clearer and the values are more accurate, a better estimation can be made on the requirements of the skin. The electrodes that are used are also made with the assumption that their resistance can be dismissed. In reality, this can be a safe assumption since the impedance of the skin is significantly higher. Well suited electrodes for t-VNS have to be designed to both minimize the voltage requirement of the circuit but also make it suitable for wearing for longer periods of time. Most of the choices in the design process took this impedance model into account.

Although several studies have illustrated that t-VNS is an effective method for epileptic treatment with certain stimulation parameters, the optimal stimulation parameters are still unknown. More research needs to be done regarding the optimal stimulation current, frequency and pulse width. However, the designed circuit has the capability to stimulate with different stimulation parameters for future testing purposes.

For the designed circuit, size is an issue due to all testing capabilities. The size of the stimulator can be decreased in later versions by removing all testing capabilities going in the direction of a final product. Size can be reduced further by fabricating the designed integrated circuit and integrating it with the electrical circuit or by combining multiple MOSFETs into one integrated circuit.

Furthermore, power consumption can be reduced by replacing the diode of the boost converter with a MOSFET to bypass the voltage drop of 0.7 V over the diode. To improve the efficiency of the circuit even further, the option to disable the PWM signal to the boost converter can be explored. It is known by the FPGA when a no load situation occurs. The capacitor will stay charged when the switch stays open. This way, no load power dissipation can be avoided. It was also observed that for higher duty cycles, the zener diode started to dissipate energy because of the high output voltage of the boost converter. These costs can be avoided by selecting components in the H-bridge and current mirror that are rated for higher voltages than 80 V. Then, the zener voltage of the zener diode can be further increased which increases the conduction threshold. This makes the design more efficient and could be the solution for the specific case of the 8 mA drive current and 9.6 V input at maximum load that could not be achieved with the current design.

In the stimulator, passive charge balancing has been applied meaning that the load will be shorted to GND to prevent charge buildup resulting in an imperfect biphasic square wave. Instead, active charge balancing could be applied to improve the stimulation wave form. When applying active charge balancing, the stimulation current is regulated in a way that expected deviations are compensated for resulting in a more perfect biphasic square wave.

## Appendix A

# Background

A.1 Project Overview

![](_page_34_Figure_3.jpeg)

Figure A.1: Project groups overview

### A.2 Impedance Model & Electrodes

![](_page_34_Figure_6.jpeg)

Figure A.2: Connection of the system

#### MONTAGUE IMPEDANCE MODEL

![](_page_35_Figure_2.jpeg)

![](_page_35_Figure_3.jpeg)

Figure A.3: Montague Impedance Model

Figure A.4: Non-linear Impedance Model

	Low	Average	High
$R_{elec} [k\Omega]$	10	10	10
$C_{elec}$ [uF]	4.6	4.6	4.6
$R_{epi} [k\Omega]$	0.88	2.27	3.66
$C_{epi}$ [nF]	48.8	28.28	7.63
$R_{tissue} [k\Omega]$	0.4	0.67	0.89

Table A.1: Table with values for Figure A.7 and Figure 2.1

Equation for calculating the resistance (R) in  $\Omega$  where  $\rho$  is the resistivity in  $\Omega \cdot m$ , l is the length in m and A is the cross section area in  $m^2$ :

$$R = \rho \frac{\iota}{A} \tag{A.1}$$

Equation for calculating the capacitance (C) in F where  $\epsilon$  is the permittivity in F/m, 1 is the distance of separation between the plates in m and A is the area of the plates in  $m^2$ :

$$C = \epsilon \frac{A}{l} \tag{A.2}$$

Table A.2: Values used for calculating  $R_{elec}$  and  $C_{elec}$  with Equation A.2 and Equation A.1

Variable	Value used
$\epsilon$	44 nF/mm
$\rho$	1kΩ/mm
Α	$1 \ cm^2$
1	0.1 cm [37]

![](_page_36_Figure_1.jpeg)

#### WET ELECTRODE-SKIN IMPEDANCE

![](_page_36_Figure_3.jpeg)

![](_page_36_Figure_4.jpeg)

Figure A.6: Electrode-skin impedance model for dry electrode

I <sub>load</sub> [mA]	V <sub>load</sub> [V] best case	V <sub>load</sub> [V] general case	V <sub>load</sub> [V] worst case	
1	2.2	5.0	8.2	
2	4.3	10.0	16.4	
3	6.5	15.1	24.7	
4	8.6	20.1	32.9	
5	10.8	25.1	41.1	
6	13.0	30.1	49.3	
7	15.1	35.2	57.5	
8	17.3	40.2	65.8	

Table A.3: Corresponding voltage over the load to drive current

#### OLD FINAL IMPEDANCE MODEL

![](_page_37_Figure_2.jpeg)

Figure A.7: Old model on which values were based

## Appendix B

# **Circuit Design**

Current Amplitude [mA]	Power to load [mW]	Power supplied [mW]	Efficiency [%]
1	2.76	34.38	8.0
2	12.43	74.70	16.64
3	29.01	115.04	25.22
4	52.12	154.83	33.66
5	82.12	194.86	42.14
6	120.00	236.00	50.85
7	163.18	275.60	59.21
8	212.83	315.12	67.54
8.8 (MAX)	261.89	351.73	74.46

### B.1 Stimulation Source

Table B.1: Average power of the current source during 1 stimulation cycle

Desired I [mA]	Measured I [mA]	Error [%]
1	1.003	0.3
2	2.004	0.2
3	3.004	0.13
4	4.001	0.025
5	4.996	-0.08
6	5.989	-0.18
7	6.98	-0.29
8	7.965	-0.4375

Table B.2: Table of measured currents with respect to desired current

|--|

Component	Value/name
NPN	2N222AUB
OP AMP	LT1490ACD
Reference resistor	681 Ω
PMOS for Mirror	SH8M41TB1

#### H-Bridge B.2

Table B.4: Com	ponents for n-bridge
Component	Value/name
BJT	2SCR514RTL
NMOS	BST82,215
Diode	STPS2H100AFY
R1,R2,R3,R4	10 kΩ
R5,R6,R7,R8	500 kΩ
R9,R10	68 Ω
C1,C2	100 nF

Current trough the load at 8 mA stimulation 108 6 4 $I_{load} \; [\mathrm{mA}]$  $\mathbf{2}$ 0 -2 -4 -6 -8 -10 L 0.1 0.3 0.4 0.6 0.7 0.8 0.9 0.20.51 t [ms]

Figure B.1: Current through the load at 8 mA stimulation

#### r h-brid **L**1 **B** 4 C

### B.3 Current Sense

	_		_	
Table B.5:	Component	values for	the curren	t sensor

Component	Value
Op-amp	MCP601
Rs	10 Ω
R1	90.9k Ω
R2	$10.1 \mathrm{k} \Omega$

	Table B.6: Simu	lation results of Current Sense		
Drive current [mA]	Output voltage [mV]	amplifier source current [uA]	$V_{Rs}$ [mV]	Power dissipation [mW]
10	998.0	227	100	2.1
9	898.0	226	90	1.9
8	798.0	225	80	1.8
7	698.0	224	70	1.6
6	598.0	223	60	1.5
5	498.0	222	50	1.4
4	398.0	221	40	1.3
3	298.0	220	30	1.2
2	198.1	219	20	1.1
1	98.1	218	10	1.1

Table B.6: Simulation results of Current Sense

### B.4 Buck Converter

	<u> </u>
Component	Value
C_in	100uF
C_out	1000uF
L	100uH
D	SD2010S040S3R0

#### B.5 Boost Converter

#### B.5.1 Component Values Calculation

According to [38], the output voltage, the critical inductance and the critical capacitance are given by formulas B.1, B.4 and B.5. The output voltage can be rewritten as equation B.3 to calculate the required duty cycle for the desired voltage output. Assuming a diode voltage of 0.6 V, a duty cycle of 0.864 is needed to step the minimum battery voltage up to 70 V and 0.822 to step the maximum battery voltage up to 70 V.

$$V_o = \frac{V_{in}}{1 - D} - V_{diode} \tag{B.1}$$

For the design it is important to determine the switching frequency. Boost converters with higher frequencies have less switching losses and can hold smaller components. However, in this application, high switching frequencies come at the cost of a lower duty cycle control resolution. The FPGA which is used to generate the PWM signal has an internal clock of 12 MHz. The resolution of the duty cycle is given by equation B.2. A switching frequency of 600 kHz brings a good balance between the size of the components and the duty cycle resolution of 5%.

$$Resolution_D = \frac{f_{switch}}{f_{clock}} * 100\%$$
(B.2)

$$D = 1 - \frac{V_{in}}{V_o + V_{diode}} \tag{B.3}$$

To calculate the minimum inductance needed, the largest critical inductance within the specification range should be derived. According to equation B.4 this value can be found by optimizing  $D(1-D)^2$  and filling in the largest possible load value. Optimizing the value of  $D(1-D)^2$  yields a value of  $\frac{1}{3}$  for the duty cycle. The maximum load occurs at 70 V and 8mA and is equal to 8750  $\Omega$ . Plugging in these values yields a minimum inductance of 1.1 mH.

$$L_C = \frac{RD(1-D)^2}{2Fs}$$
(B.4)

The minimum value for the capacitance can be derived by using equation B.5. This value can be found by optimizing  $\frac{V_o}{R}$  which occurs at the largest possible output current which is 8 mA in this application.  $V_o$  is at most 70 V and the allowed ripple voltage is chosen to be at most 100 mV since there is no hard requirement but does keep the voltage steady. Plugging in these values together with the switching frequency of 600 kHz and the maximum duty cycle of 0.864 yields a minimum value of 112 nF for the capacitor

$$C_B = \frac{V_o D}{F s \Delta V_o R} \tag{B.5}$$

The diode that is selected for the design is a Schottky diode because of its low forward voltage drop. The diode should be able to handle current peaks of 490 mA. The electrical switch that is selected for the design is an N-Channel MOSFET with a very low gate charge for fast switching. The MOSFET can be directly controlled from the FPGA at 600 kHz frequency. Lastly, a zener diode is needed to fulfill the requirement regarding 0 mA output current. Actively switching boost converters require a load for proper functionality. If no load is connected, the capacitor will rise until the circuit breaks. A zener diode functions as a load above a certain specified voltage(zener voltage) to prevent overvoltage by only dissipating power when needed. Due to the maximum drain to source voltage specifications of 80 V in the current mirror, a 75 V zener diode is chosen.

<b>DIC B.O. BOOD</b>	eonverter component n
Component	Value
L	2.1mH
С	180nF
Q	IRLML0100TRPBF
D	STPS2H100AFY

Table B.8: Boost converter component list

### B.5.2 Simulation results

![](_page_43_Figure_4.jpeg)

Figure B.2: Example output voltage of a soft start sequence

Table B.9: Simulated efficiencies [%] as a function of drive current [mA] for a general case load and a 9.6V battery

Current [mA]	Duty cycle [%]	Input Power [mW]	Output efficiency [%]	Load efficiency [%]
1	0	9.6	96.9	52.1
2	5	38.4	66.7	52.1
3	40	69.1	77.9	65.5
4	50	104.6	78.3	76.8
5	60	158.4	79.1	79.2
6	65	241.9	80.9	74.7
7	70	325.4	80.7	75.7
8	75	439.7	81.0	73.1

	off	12.3	12.2	12.2	12.2	12.2	12.2	12.2	12.2
	85%	74.6	74.6	74.6	74.6	74.6	74.6	74.4	74.3
tage	80%	78.4	71.3	66.2	62.7	60.2	58.5	57.1	55.9
iput vol	75%	62	55.8	51.9	49.3	47.6	46.3	45.3	44.5
9.6V ii	70%	50.9	45.6	42.6	40.6	39.3	38.3	37.5	36.9
‰] for a	65%	42.9	38.5	36	34.5	33.4	32.6	31.9	31.5
cycle [9	60%	37.5	33.3	31.2	29.9	29	28.3	27.8	27.4
an duty	55%	32.8	29.2	27.4	26.3	25.6	25	24.6	24.3
t [mA]	50%	29	26	24.5	23.5	22.9	22.4	22	21.7
curren	45%	25.7	23.2	21.9	21	20.5	20	19.7	19.5
of drive	40%	23.3	21	19.7	19	18.5	18.1	17.9	17.6
unction	35%	21.2	19.1	18	17.3	16.9	16.6	16.3	16.1
/] as a f	30%	19.6	17.6	16.5	15.9	15.5	15.2	15	14.8
ltage [V	25%	18.2	16.3	15.3	14.8	14.4	14.1	13.9	13.7
atput vo	20%	17	15.2	14.3	13.7	13.4	13.1	12.9	12.8
ilated oi	15%	16.1	14.3	13.4	12.9	12.5	12.3	12.1	11.9
0: Simu	10%	15.3	13.4	12.6	12.1	11.8	11.5	11.4	11.2
ble B.1	5%	14.6	12.8	11.9	11.4	11.1	10.9	10.7	11.2
Ta	Current [mA] / D	1	2	3	4	5	9	7	8

er diode current (peak [mA], average peak [mA], average steady state [mA]) as a function of drive current [mA]
er diode current (peak [mA], average peak [mA], average steady state [mA]) as a function of driv
er diode current (peak [mA], average peak [mA], average steady state [mA]) as a f
er diode current (peak [mA], average peak [mA], average steady sta
er diode current (peak [mA], average peak [mA], aver
er diode current (peak [mA], average pe
er diode current (peak [m
er diode cu
zen
alated output a put voltage
able B.11: Sim [8] for a 9.6V in

off	0	0	0	0	0	0	0	0
85%	320 40 10	320 40 8	320 40 6	320 40 4	320 40 4	320400	320400	320400
80%	47 15 0.5	45 15 0	45 15 0	39 15 0	35 14 0	32 13 0	27 12 0	23 10 0
75%	0	0	0	0	0	0	0	0
20%	0	0	0	0	0	0	0	0
65%	0	0	0	0	0	0	0	0
60%	0	0	0	0	0	0	0	0
55%	0	0	0	0	0	0	0	0
50%	0	0	0	0	0	0	0	0
45%	0	0	0	0	0	0	0	0
40%	0	0	0	0	0	0	0	0
35%	0	0	0	0	0	0	0	0
30%	0	0	0	0	0	0	0	0
25%	0	0	0	0	0	0	0	0
20%	0	0	0	0	0	0	0	0
15%	0	0	0	0	0	0	0	0
10%	0	0	0	0	0	0	0	0
5%	0	0	0	0	0	0	0	0
Current [mA] / D	1	2	3	4	5	9	7	8

	off	0	0	0
	%08	259.2	428.46	811.44
ad case	75%	163.2	202.02	294.84
or a no lo	20%	121.92	150.96	181.44
tage [V] f	65%	95.04	117.66	139.86
l input vol	60%	77.76	93.24	114.66
cycle and	55%	66.24	L'LL	91.98
of duty o	50%	55.68	68.82	81.9
function	45%	49.92	58.83	69.3
nW] as a	40%	44.16	54.39	64.26
aption [n	35%	42.24	52.17	60.48
r consun	30%	38.4	49.95	59.22
ter powe	25%	38.4	45.51	57.96
st convei	20%	36.48	45.51	55.44
ated Boo	15%	35.52	46.62	55.44
2: Simula	10%	39.36	45.51	57.96
able B.12	5%	38.4	48.84	59.22
T	Input [V] / D	9.6	11.1	12.6

#### **B.5 Boost Converter**

44

	D.1.0.	JIIIUIau	Tu pear	Indino	VUILABO			WOV	224		20 J J	2000	200	E C C C	E L	2000	2010	J
Current [mA] / D	%c	10%	15%	20%	0%C7	30%0	35%	40%	45%	%0C	%cc	0//0	%C0	/0%	0%C1	80%	%C8	ОĦ
1	x	x	x	x	x	х	x	x	x	x	x	38	44.1	52	62.6	х	74.7	x
2	x	x	x	x	x	18	20.1	22.5	25.4	29	32.9	37.7	43.7	51.6	62.2	LL	74.7	x
3	x	x	13.6	14.7	16.1	17.8	19.9	22.3	25.2	28.7	32.6	37.4	43.4	51.3	61.8	76.6	74.7	x
4	11.5	12.40	13.4	14.6	16	17.7	19.7	22.1	25	28.5	32.4	37.2	43.2	50.9	61.4	76	74.7	x
5	11.4	12.3	13.2	14.4	15.9	17.5	19.6	22	24.9	28.3	32.2	36.9	42.7	50.5	60.9	75.5	74.8	x
9	11.3	12.2	13.2	14.3	15.8	17.4	19.5	21.8	24.6	28.1	31.9	36.6	42.5	50.3	60.5	75	74.7	x
7	11.2	12.1	13	14.2	15.6	17.3	19.3	21.6	24.5	27.9	31.7	36.4	42.2	49.7	60.2	74.6	74.7	x
8	12.1	12	12.9	14.2	15.6	17.2	19.2	21.5	24.3	27.7	31.5	36.1	37.8	49.6	59.7	74.1	74.7	x
Та	ble B.1	4: Simul	lated inp	out curre	ant [mA]	] as a fu	inction (	of drive	current	: [mA] a	in duty (	cycle [%	6] for a	9.6V in	put volt	age		
Current [mA] / D	5%	10%	15%	20%	25%	30%	35%	40%	45%	50%	55%	60%	65%	70%	75%	80%	85%	off
1	2.4	2.5	2.6	2.9	3.4	3.5	4	4.5	5.2	9	6.9	8.3	10.3	13.1	17.8	25.5	84.1	-
2	4	4.3	4	4.2	4.5	5	5.5	6.1	7	8.1	9.4	10.8	13.3	16.4	20.8	28.8	84.1	5
3	4.6	4.8	5	5.5	5.9	6.5	7.2	7.9	8.9	10.1	11.7	13.4	16.4	19.6	24.9	33.1	84.3	ю
4	5.6	9	6.3	6.7	7.4	~	8.8	9.7	10.9	12.5	14.1	16.3	19.2	23.5	29.3	38.2	84.5	4
l	\$	, 1	ı t	, ,	L C	-			1	L		-						l

	off	1	6	e	4	S	9	٢	$\infty$
	85%	84.1	84.1	84.3	84.5	85.7	86.7	84.9	86
3	80%	25.5	28.8	33.1	38.2	44	48.9	54.4	59.4
-	75%	17.8	20.8	24.9	29.3	33.3	37.7	41.6	45.8
	70%	13.1	16.4	19.6	23.5	26.6	30.1	33.9	37.3
•	65%	10.3	13.3	16.4	19.2	22.3	25.2	28.3	31.4
, ,	60%	8.3	10.8	13.4	16.3	19.1	21.7	24.4	26.9
r	55%	6.9	9.4	11.7	14.1	16.5	18.7	21.4	23.6
-	50%	9	8.1	10.1	12.5	14.5	16.9	19	21.1
	45%	5.2	7	8.9	10.9	12.7	14.8	16.8	18.7
	40%	4.5	6.1	7.9	9.7	11.6	13.2	15	16.7
	35%	4	5.5	7.2	8.8	10.5	12	13.5	15.1
-	30%	3.5	5	6.5	8	9.4	10.9	12.4	13.9
	25%	3.4	4.5	5.9	7.4	8.5	10.1	11.5	13
	20%	2.9	4.2	5.5	6.7	8.1	9.4	10.7	11.9
-	15%	2.6	4	5	6.3	7.5	8.8	9.9	11.1
	10%	2.5	4.3	4.8	9	7.1	8.2	9.3	10.5
	5%	2.4	4	4.6	5.6	6.8	7.8	9.1	10.5
	Current [mA] / D	1	2	3	4	5	9	7	8

e	
oltag	
ţ	
ndt	
∕ ir	
6	
a 9	
or	
<u>[</u> ]	
6	
cle	
сy	
uty	
ղ	
ar	
[A	
[]	
ent	
Ĕ	
อ	
ive	
[ di	
10	
E OI	
nci	
Ę	
ls a	
5	
<u>~</u>	
lcy	
ier	
ΕŬ	
d e	
ate	
ulí.	
Sin	
.1.	
B	
lblé	
Ę	

	off	97.6	96.8	96.8	96.8	96.8	96.8	96.8	96.8
	85%	9.2	18.5	27.7	36.8	45.3	53.8	63.9	72.0
5 G	80%	32.0	51.6	62.5	68.4	71.3	74.8	76.5	78.4
ut voltag	75%	36.3	55.9	65.1	70.1	74.4	76.8	79.4	81.0
oV Inpi	70%	40.5	57.9	67.9	72.0	77.0	79.5	80.7	82.4
tor a 9.	65%	43.4	60.3	68.6	74.9	78.0	80.9	82.2	83.6
'cle [%]	60%	47.1	64.2	72.8	76.4	79.1	81.5	83.1	84.9
duty cy	55%	49.5	64.7	73.2	77.7	80.8	83.6	83.8	85.8
mA] an	50%	50.3	60.9	75.8	78.3	82.3	82.8	84.4	85.7
urrent	45%	51.5	69.0	76.9	80.3	84.1	84.5	85.5	86.9
drive c	40%	53.9	71.7	9.77	81.6	83.1	85.7	87.0	87.8
iction of	35%	55.2	72.3	78.1	81.9	83.8	86.5	88.0	88.9
as a tur	30%	58.3	73.3	79.3	82.8	85.9	87.2	88.2	88.7
ICY [%]	25%	55.8	75.5	81.0	83.3	88.2	87.3	88.1	87.8
emcier	20%	61.1	75.4	81.3	85.2	86.2	87.1	87.9	89.6
mulated	15%	64.5	74.5	83.8	85.3	86.8	87.4	89.1	89.3
IIS :CI .	10%	63.8	64.9	82.0	84.0	86.6	87.7	89.4	88.9
Table E	5%	63.4	66.7	80.8	84.8	85.0	87.3	85.7	88.9
	Current [mA] / D	1	2	3	4	5	9	7	8

Current [mA] / Duty cycle	80%	85%
1	74.7	74.7
2	74.7	74.8
3	74.6	74.8
4	74.6	74.7
5	74.6	74.8
6	74.6	74.7
7	74.6	74.8
8	74.5	74.8

Table B.16: Simulated output voltage [V] as a function of drive current [mA] an duty cycle [%] for a 12.6V input voltage

Table B.17: Simulated output zener diode current (peak [mA], average peak [mA], average steady state [mA]) as a function of drive current [mA] and uty cycle [%] for a 12.6V input voltage

Current [mA] / Duty cycle	80%	85%
1	360 60 9	550 x 70
2	360 60 7	550 x 70
3	360 60 6	550 x 55
4	360 60 5	550 x 73
5	360 60 4	550 x 60
6	360 60 3	550 x 57
7	360 60 2	550 x 55
8	360 60 1	550 x 58

Table B.18: Simulated peak output voltage [V] as a function of drive current [mA] an duty cycle [%] for a 12.6V input voltage

Current [mA] / Duty cycle	80%	85%
1	74.8	74.8
2	74.8	74.8
3	74.8	74.8
4	74.8	74.8
5	74.7	74.8
6	74.7	74.9
7	74.9	74.8
8	74.7	74.9

Current [mA] / Duty cycle	80%	85%
1	64.2	394
2	64.3	417
3	64.2	417
4	64.4	404
5	64.6	400
6	64.7	396
7	64.9	394
8	65.2	400

Table B.19: Simulated input current [mA] as a function of drive current [mA] an duty cycle [%] for a 12.6V input voltage

Table B.20: Simulated efficiency [%] as a function of drive current [mA] an duty cycle [%] for a 12.6V input voltage

Current [mA] / Duty cycle	80%	85%
1	12.1	2.0
2	24.2	3.7
3	36.3	5.6
4	48.3	7.7
5	60.1	9.7
6	72.1	11.8
7	83.8	13.8
8	95.2	15.6

Table B.21: Simulated Zener current [mA] and power [mW] as a function of duty cycle and input voltage [V] for a no load case

Input [V]	Current [mA] (80% Duty cycle)	Current [mA] (75% Duty cycle)	Current [mA] ( <75% Duty cycle)
9.6	1.5	0	0
11.1	3.6	0.8	0
12.6	8.3	1.9	0
Input [V]	Power [mW] (80% Duty cycle)	Power [mW] (75% Duty cycle)	Power [mW] (<75% Duty cycle)
Input [V] 9.6	Power [mW] (80% Duty cycle) 112.5	Power [mW] (75% Duty cycle) 0	Power [mW] (<75% Duty cycle) 0
Input [V] 9.6 11.1	Power [mW] (80% Duty cycle) 112.5 270	Power [mW] (75% Duty cycle) 0 60	Power [mW] (<75% Duty cycle) 0 0

# Appendix C

## Integrated Circuit

The sub-modules of the integrated circuit will be explained in more detail in the following sections. More information about the signals displayed in the figures can be found in section C.7.

### C.1 Decoder

The decoder is responsible for translating the messages from the microcontroller to appropriate signals. The messages contain the stimulation parameters that are forwarded to their respective components after their decoding. The decoder receives the following signals from the microcontroller:

- Stimulation intensity ranging from 0 to 8 mA. For this, 3 bits are enough to read the signal.
- Stimulation frequency ranging from 10 Hz to 40 Hz in steps of 5 Hz. This will again require 3 bits to be send to determine the correct frequency.

The frequency needs to be converted to a time value since it will be directly passed through a timer. The formula for converting a frequency is:  $T = \frac{1}{t}$ .

![](_page_48_Figure_8.jpeg)

Figure C.1: Decoder diagram

The look-up tables are made to decode the signals accordingly and they can be found in Table C.3. The bit column indicates what options there are with 3 bits, the  $I_{amp}$  column indicates what those values would mean for the current and the  $T_{ip}$  column indicates the pulse times that are required.  $T_{ip}$  is the time between the biphasic stimulation pulse and the next start of the following pulse.

The  $T_{ip}$  needs a subtraction of two times the pulse time for each one. This is because as indicated in the figure in Figure 2.2, first two pulses have passed of each  $200\mu s$ . Therefore this still has to be taken into

	÷	
Bits	$I_{amp}$ [mA]	Tip [ms]
000	1	100
001	2	67
010	3	50
011	4	40
100	5	33
101	6	29
110	7	25
111	8	No Stimulation
	Bits 000 001 010 011 100 101 110 111	$\begin{tabular}{ c c c c c c c c c c c c c c c c c c c$

Table C.1: Decoding of the corresponding bits for each important signal

account in the form of removing the time that has already elapsed from the time in Table C.1.

The  $D_{max}$  is a value that is required for the boost converter to have optimal power consumption. More info can be found in the section that covers boost control in section C.5. The decoding table can be found in Table C.2.

Table C.2: Decoding of the reference current to the minimum reference and maximum reference voltage

$I_{amp}$ [mA]	$D_{max}$ [clock cyles]	$D_{max}$ [%]
1	0	0
2	6	30
3	11	55
4	13	65
5	15	75
6	16	80
7	16	80
8	16	80

The other use case that is specified is a simple turn off the circuit signal. When this signal is received from the microcontroller it will be considered as a faulty signal and then the circuit will power down.  $V_{max}$  and  $V_{min}$  reference signals are the signals from which undercurrent and overcurrent are determined. The decoding table for this can be found in Table C.3

Table C.3: Decoding of the reference current to the minimum reference and maximum reference voltage

$I_{amp}$ [mA]	$V_{max}$ [V]	$V_{min}$ [V]
1	0.12	0.08
2	0.22	0.18
3	0.32	0.28
4	0.42	0.38
5	0.52	0.48
6	0.62	0.58
7	0.72	0.68
8	0.82	0.78

### C.2 H-bridge Control

The H-bridge control sub-module with its corresponding in- and outputs is shown in Figure C.2.

![](_page_50_Figure_3.jpeg)

Figure C.2: H-bridge Diagram

It has predetermined times (the pulse width) and times provided from the outside (inter-pulse time). The reason that the H-bridge control has its own timer is because it has many states that it has to keep track of. These all progress automatically based on if a pre-indicated time has passed. This is the time\_goal that is passed from the H-bridge control into the timer.

The reason that the control has so many states is because of the relation it has with the current source. The first reason is that the current source should start up. This is done by shorting the gates of the H-bridge but leaving one gate open. As a result, only one other gate has to be opened to let the current. When the current is flowing and the corresponding lower gate is opened, it can start pushing current through the load. When the time has passed of the first pulse, again a short circuit is done for a brief period of time. This is implemented because otherwise high voltage spikes would occur which is very undesired given the load that is being used. Then the other pulse is send for the time of 1 pulse width and after that another pause to make sure that no more current flows through the load. Then the current source is turned off and the H-bridge is placed in the inter-pulse time in which the load is discharged. When this time has passed, the cycle starts over again. This process can be seen in Figure C.5.

During some of these periods, the current through the sensing resistor becomes quite large due to the short circuits that occur. These are important moments because that is when the measure\_enable signal is turned off. The reason for this is because the current that is being measured is simply not going through the load. Therefore, this signal is important to turn off when the short circuit occurs.

![](_page_51_Figure_1.jpeg)

Figure C.3: H-bridge FSM

![](_page_52_Figure_1.jpeg)

Figure C.4: H-bridge-timer FSM

![](_page_52_Figure_3.jpeg)

![](_page_52_Figure_4.jpeg)

![](_page_52_Figure_5.jpeg)

### C.3 Current Sense Control

![](_page_53_Figure_2.jpeg)

Figure C.6: Current sense diagram

The current sense control in Figure C.6 is responsible for the processing of the voltage that is measured over the sensing resistor. This voltage will be compared with both the maximum and the minimum reference voltage. When the voltage that is sensed is higher than the maximum reference voltage, this means that the current through the sensing resistor is too high. In this situation, the Overcurrent Indicator (OCI) signal will be high. When the voltage is lower than the minimum reference voltage, the current through the sensing resistor is lower than should be. This is a situation of undercurrent, in which the Undercurrent Indicator (UCI) signal is high. When neither is the case, it should be in between the safety margins and therefore no need for any action.

The measure\_enable signal is the signal that is received from the H-bridge. The explanation for this signal can be found in section C.2. The signal is a basic indication on whether the signal from the sensing resistor should be taken into account. The FSM can be found in Figure C.7.

![](_page_53_Figure_6.jpeg)

Figure C.7: Current Sense FSM

### C.4 DAC

As mentioned in Section 5.1, the current source requires a variable voltage source to generate the reference currents. In order to translate the binary value of the stimulation current into the required input voltage of the current source an DAC is used. For this the MAX5513 [39] was chosen, which is a power efficient DAC with an 8-bit resolution with respect to the reference voltage. To drive the DAC with the integrated circuit, the DAC communication sub-module is required. The DAC expects a serial input including control bits, data bits and sub-bits. In addition, it expects the serial communication clk and a signal which indicates that a command is being sent. An overview of the sub-module with its corresponding in- and outputs is shown in Figure C.8.

![](_page_54_Figure_3.jpeg)

Figure C.8: Overview of the DAC sub-module

The sub-module expects an 3-bit binary number for the required stimulation current (I\_Load) and an 1bit signal which enables the serial communication (DAC\_Enable). The Look Up Table translates the 3-bit binary value for the stimulation current into the required 8 data bits. Furthermore, the Look Up Table sends the required control bits. The required timing of the serial communication is shown in Figure C.9.

![](_page_54_Figure_6.jpeg)

Figure C.9: DAC register loading diagram [39]

The DAC Parallel to Serial module consists of an FSM for sending the serial communication. The FSM is shown in Figure C.10. It consists of 18 states. One reset state, one enable state and then 16 states for sending all bits.

![](_page_55_Figure_1.jpeg)

Figure C.10: DAC communication FSM

#### C.5 Boost Control

The boost control module actively controls the voltage output of the boost converter by changing the PWM signal of it's N-channel MOSFET. Information about the effect of changing about the PWM signal can be found in section 5.5. The sub-module decomposition of the boost control module can be found in figure C.16.

![](_page_56_Figure_3.jpeg)

Figure C.11: The Boost control sub-module overview

As described in section 5.5, the boost converter output voltage needs to match the load for the highest efficiency. An algorithm has been implemented in the Duty cycle Control sub-module to find the corresponding duty cycle. This algorithm works as follows: Initially the duty cycle is stepped up gradually until it reaches the maximum allowed duty cycle indicated by the decoder (which depends on the drive current). Then, the duty cycle is slowly stepped down after every stimulation pulse until undercurrent is detected. The undercurrent data storage sub-module makes sure that every undercurrent event will be detected by Duty cycle control. When undercurrent is detected, the Duty cycle control sets its value to last duty cycle is maintained. Another undercurrent event will reset the algorithm. With the Duty cycle and a counter, PWM Control will create the desired signal. To avoid an excessive amount of arrows in the duty cycle control FSM, one conditions has been skipped in the figure. All states will have the condition that when that when the 'Enable' signal becomes a logical 0, the Reset state will be entered during the next clock cycle.

![](_page_57_Figure_1.jpeg)

Figure C.12: Duty Cycle control FSM

![](_page_57_Figure_3.jpeg)

Figure C.13: Undercurrent Data Storage FSM

![](_page_57_Figure_5.jpeg)

Figure C.14: Soft Start Counter FSM

![](_page_58_Figure_1.jpeg)

Figure C.15: PWM Control FSM

![](_page_58_Figure_3.jpeg)

Figure C.16: PWM Counter

### C.6 Main Control

![](_page_59_Figure_2.jpeg)

Figure C.17: The Main Control unit with inputs and outputs

The main control unit is responsible for processing the information that is externally received from the microcontroller through the decoder and for responding to signals involving the components. Although it is a very important element in itself, it does not consist of a lot of difficult logic. Its main purpose is to respond to overcurrent from the current sense or a message from the boost converter specifying a situation of undercurrent and the need to up the load. The main control unit is the only way for other control units to communicate with the DAC. Other than listening to those parameters, it is also responsible for turning on other control units such as the H-bridge and the boost converter. The FSM can be found in Figure C.18 and the accompanying FSM in Figure C.19.

In a situation of undercurrent, the boost converter sends a digital 1 to the main control. This pulse is read and is checked with the current undercurrent counter. The undercurrent counter is a measure to check if there are too many consecutive undercurrent pulses. Too much undercurrent means that the impedance of the load is too high which could be due to the fact that the electrodes are poorly connected. As a threshold, receiving 5 undercurrent pulses in a row would indicate that something is wrong which would put the main control in a different state. From this state, a signal is send back to the microcontroller to indicate that something went wrong and that the device needs to be shut down. It is important that the device is shutdown manually because it is important to follow the startup process for the boost converter to build up voltage.

![](_page_60_Figure_1.jpeg)

Figure C.18: Main Control FSM

![](_page_60_Figure_3.jpeg)

Figure C.19: Control counter for Main Control

### C.7 Signal Declarations

Signal Name	Bits	Explanation
Boost_Done	1	Signal which indicates whether or not the boost startup sequence is completed
Boost_Enable	1	On / off signal for the Boost conrol
Boost_PWM	1	PWM output connected to boost converter
Control	4	DAC communication settings
Current_enable	1	Indication that the H-bridge is in a state in which it can conduct current
DAC_enable	1	Signal to enable or disable current source
D	5	Duty cycle counter value for PWM generation
D_max	5	Maximum allowed duty cycle
DAC_SLCK	1	1 out of 3 DAC communication signals
DAC_Write	1	1 out of 3 DAC communication signals
Data	8	Data to be transmitted to DAC
Fault	1	Error flag which can be read by the microcontroller
H_bridge_enable	1	On / off signal for the H-bridge
I_amp	3	Current magnitude to stimulate with (0 to 8)
I_load	3	Signal to set the current of the current source
Measure_Enable	1	Signal that indicates when accurate current measurements can be taken
Microcontroller_in	6	Encoded stimulation current and frequency
OCI	1	Over Current Indicator,
Pulse_End	1	End of a biphasic pulse indicator.
PWM_C_Reset	1	digital 1 Resets the PWM counter, 0 allows counter to run
PWM_C_Value	5	Value of the PWM counter
Serial_DAC	1	1 out of 3 DAC communication signals
Soft_C_Value	14	Counter value from soft start counter - Boost control
Soft_C_Reset	1	Reset signal for soft start counter - Boost control
Startup	1	Signal indicating startup sequence of boost control
Status	1	Signal which indicates wether or not stimulation is needed
Stored_UCI	1	Stored Under Current Indicator Value. Remembered for 1 stimulation period
s1	1	H Bridge transistor signal 1
s2	1	H Bridge transistor signal 2
s3	1	H Bridge transistor signal 3
s4	1	H Bridge transistor signal 4
Tip	21	Timer value for interpulse time
time	21	timer value - H-bridge control
time_goal	21	The time for which a given state should last inside the H-bridge control
UCI	1	Under Current Indicator
UCI_Pulse	1	Under Current Indicator Pulse
V_Max_Ref	12	Maximum Allowed value for the ADC (to detect overcurrent)
V_Min_Ref	12	Minimum Allowed value for the ADC (to detect undercurrent)
V_sense	12	ADC value

Table C.4: IC Signal Declarations

## Bibliography

- [1] WHO, "Epilepsy: a public health imperative," 2019, June 13. https://www.who.int/publications/i/item/epilepsy-a-public-health-imperative.
- [2] LivaNova, "Sentiva Duo."
- [3] P. Boon, E. D. Cock, A. Mertens, and E. Trinka, "Neurostimulation for drug-resistant epilepsy: a systematic review of clinical evidence for efficacy, safety, contraindications and predictors for response," *Current Opinion in Neurology*, vol. 31, pp. 198–210, 2018.
- [4] H. Stefan, G. Kreiselmeyer, F. Kerling, K. Kurzbuch, C. Rauch, M. Heers, B. S. Kasper, T. Hammen, M. Rzonsa, E. Pauli, J. Ellrich, W. Graf, and R. Hopfengärtner, "Transcutaneous vagus nerve stimulation (t-vns) in pharmacoresistant epilepsies: A proof of concept trial," *Epilepsia*, vol. 52, pp. 115–118, 2012.
- [5] P. Rong, A. Liu, J. Zhang, Y. Wang, W. He, A. Yang, L. Li, H. Ben, L. Li, H. Liu, P. Wu, R. Liu, Y. Zhao, J. Zhang, F. Huang, X. Li, and B. Zhu, "Transcutaneous vagus nerve stimulation for refractory epilepsy: a randomized controlled trial," *Clinical Science*, 2014.
- [6] J. Kong, J. Fang, J. Park, S. Li, and P. Rong, "Treating depression with transcutaneous auricular vagus nerve stimulation: State of the art and future perspectives," *Frontiers in Psychiatry*, vol. 9, 2018.
- [7] L. E. W. W. S. P. K. T. Yap JYY, Keatch C, "Critical review of transcutaneous vagus nerve stimulation: Challenges for translation to clinical practice," *Frontiers in neuroscience*, vol. 14, no. 284, 2020.
- [8] Y. Yu, J. Ling, L. Yu, P. Liu, and M. Jiang, "Closed-loop transcutaneous auricular vagal nerve stimulation: Current situation and future possibilities," *Frontiers in Human Neuroscience*, vol. 15, 2022.
- [9] A. Coston and J.-J. Li, "Transdermal drug delivery: a comparative analysis of skin impedance models and parameters," in *Proceedings of the 25th Annual International Conference of the IEEE Engineering in Medicine and Biology Society (IEEE Cat. No.03CH37439)*, vol. 3, pp. 2982–2985 Vol.3, 2003.
- [10] J. Simatupang, W. Wijaya, D. Tyler, and C. Mavridis, "Simulation analysis of equivalent circuit model of skin-electrode impedance for transcutaneous electrical stimulation," *Bulletin of Electrical Engineering and Informatics*, vol. 10, pp. 1936–1943, 08 2021.
- [11] A. Boxtel, "Skin resistance during square-wave electrical pulses of 1 to 10 ma," *Medical biological engineering computing*, vol. 15, pp. 679–87, 12 1977.
- [12] S. Dorgan and R. Reilly, "Model for human skin impedance during surface functional neuromuscular stimulation," *IEEE transactions on rehabilitation engineering : a publication of the IEEE Engineering in Medicine and Biology Society*, vol. 7, pp. 341–8, 10 1999.
- [13] K. M. C. R. J. A. M. W. Vargas Luna, J. L., "Dynamic impedance model of the skin-electrode interface for transcutaneous electrical stimulation.," *PLoS One*, vol. 10, 2015.
- [14] L. Colzato and C. Beste, "A literature review on the neurophysiological underpinnings and cognitive effects of transcutaneous vagus nerve stimulation: challenges and future directions," *Journal of Neurophysiology*, vol. 123, no. 5, pp. 1739–1755, 2020. PMID: 32208895.

- [15] N. Yakunina, S. S. Kim, and E.-C. Nam, "Optimization of transcutaneous vagus nerve stimulation using functional mri," *Neuromodulation: Technology at the Neural Interface*, vol. 20, no. 3, pp. 290– 300, 2017.
- [16] M. F. Butt, A. Albusoda, A. D. Farmer, and Q. Aziz, "The anatomical basis for transcutaneous auricular vagus nerve stimulation," *Journal of Anatomy*, vol. 236, no. 4, pp. 588–611, 2020.
- [17] M. L. I. L. J. L. Iñaki Garcia de Gurtubay, Pedro Bermejo, "Evaluation of different vagus nerve stimulation anatomical targets in the ear by vagus evoked potential responses," *Brain and Behavior*, vol. 11, no. 11, 2021.
- [18] S. Andrews, W. J. Lee, and M. Prausnitz, "Recovery of skin barrier after stratum corneum removal by microdermabrasion," AAPS PharmSciTech, vol. 12, no. 4, pp. 1393–1400, 2011.
- [19] G. Li, S. Wang, and Y. Y. Duan, "Towards conductive-gel-free electrodes: Understanding the wet electrode, semi-dry electrode and dry electrode-skin interface impedance using electrochemical impedance spectroscopy fitting," *Sensors and Actuators B: Chemical*, vol. 277, pp. 250–260, 2018.
- [20] B. B. Murphy, B. H. Scheid, Q. Hendricks, N. V. Apollo, B. Litt, and F. Vitale, "Time evolution of the skin–electrode interface impedance under different skin treatments," *Sensors*, vol. 21, no. 15, 2021.
- [21] R. Butikofer and P. D. Lawrence, "Electrocutaneous nerve stimulation-i: Model and experiment," *IEEE Transactions on Biomedical Engineering*, vol. BME-25, no. 6, pp. 526–531, 1978.
- [22] R. Butikofer and P. D. Lawrence, "Electrocutaneous nerve stimulation-ii: Stimulus waveform selection," *IEEE Transactions on Biomedical Engineering*, vol. BME-26, no. 2, pp. 69–75, 1979.
- [23] L. E. W. W. S. P. K. T. Yap JYY, Keatch C, "Critical review of transcutaneous vagus nerve stimulation: Challenges for translation to clinical practice," *Frontiers in neuroscience*, vol. 14, no. 284, 2020.
- [24] S. L. Thompson, G. H. O'Leary, C. W. Austelle, E. Gruber, A. T. Kahn, A. J. Manett, B. Short, and B. W. Badran, "A review of parameter settings for invasive and non-invasive vagus nerve stimulation (vns) applied in neurological and psychiatric disorders," *Frontiers in Neuroscience*, vol. 15, 2021.
- [25] S. Eickhoff and J. C. Jarvis, "Pulse shaping strategies for electroceuticals: A comprehensive survey of the use of interphase gaps in miniature stimulation systems," *IEEE Transactions on Biomedical Engineering*, vol. 68, no. 5, pp. 1658–1667, 2021.
- [26] A. Scheiner, J. Mortimer, and U. Roessmann, "Imbalanced biphasic electrical stimulation: Muscle tissue damage," *Annals of biomedical engineering*, vol. 18, pp. 407–25, 02 1990.
- [27] S. K, T. Stieglitz, and M. Ortmanns, "An experimental study on passive charge balancing," Advances in Radio Science Kleinheubacher Berichte, vol. 7, 05 2009.
- [28] N. Nitta, F. Wu, J. T. Lee, and G. Yushin, "Li-ion battery materials: present and future," *Materials Today*, vol. 18, no. 5, pp. 252–264, 2015.
- [29] Digilent, CMOD A7 FPGA. Rev. B.
- [30] J. Vargas Luna, M. Krenn, J. Cortes-Ramirez, and W. Mayr, "Comparison of current and voltage control techniques for neuromuscular electrical stimulation in the anterior thigh," *Biomedizinische Technik. Biomedical engineering*, vol. 58, 09 2013.
- [31] S. L. Thompson, G. H. O'Leary, C. W. Austelle, E. Gruber, A. T. Kahn, A. J. Manett, B. Short, and B. W. Badran, "A review of parameter settings for invasive and non-invasive vagus nerve stimulation (vns) applied in neurological and psychiatric disorders," *Frontiers in Neuroscience*, vol. 15, 2021.
- [32] B. Aggarwal, M. Gupta, and A. Gupta, "A comparative study of various current mirror configurations: Topologies and characteristics," *Microelectronics Journal*, vol. 53, pp. 134–155, 2016.

- [33] V. Ulansky, A. Raza, and D. Milke, "Two-terminal electronic circuits with controllable linear ndr region and their applications," *Applied Sciences*, vol. 11, p. 9815, 10 2021.
- [34] B. S. Lee, "Understanding the terms and definitions of Ido voltage regulators," 1999.
- [35] Texas Instruments, Step-Down Voltage Regulator, 5 2021. Rev. 1.
- [36] On Semi, MAX810 Microprocessor Reset Monitor. Rev. B.
- [37] J. Danter, R. Siegert, and H. Weerda, "Ultrasonographische haut- und knorpeldickenmessungen an gesunden und rekonstruierten ohren mit einem 20-mhz-ultraschallgerät," *Laryngorhinootologie*, vol. 75, pp. 91–94, 1996.
- [38] B. M. Hasaneen and A. A. Elbaset Mohammed, "Design and simulation of dc/dc boost converter," in 2008 12th International Middle-East Power System Conference, pp. 335–340, 2008.
- [39] Maxim Integrated Products, Dual, Ultra-Low-Power, 8-Bit, Voltage-Output DAC, 2004.